

Contents lists available at ScienceDirect

### Global Environmental Change



journal homepage: www.elsevier.com/locate/gloenvcha

# Global energy consumption of the mineral mining industry: Exploring the historical perspective and future pathways to 2060



Emmanuel Aramendia<sup>a,\*</sup>, Paul E. Brockway<sup>a</sup>, Peter G. Taylor<sup>a,b</sup>, Jonathan Norman<sup>a</sup>

<sup>a</sup> Sustainability Research Institute, School of Earth and Environment, University of Leeds, Leeds LS2 9JT, UK

<sup>b</sup> Sustainable Systems and Processes, School of Chemical and Process Engineering, University of Leeds, Leeds LS2 9JT, UK

#### ARTICLE INFO

Keywords: Mining industry Mineral depletion Mineral requirements Energy requirements Energy transition Energy-economy models

#### ABSTRACT

The mining industry globally is responsible for significant energy consumption, and is an important source of greenhouse gas emissions. Considering that future mineral demand is likely to increase and that the final energy consumption per unit mass of mineral extracted (energy intensities of mining) is also forecast to increase as a result of a decrease in mineral resource deposit qualities, the mining industry's final energy consumption will increase in the future. But the scale of that future increase remains unexplored. In this study, we (i) provide the first bottom-up assessment of the mining industry's final energy consumption globally (1971–2015), (ii) use  $1.5^{\circ}$ C consistent socio-economic scenarios to conduct an exploratory study of future possible pathways for the mining industry's final energy consumption, and (iii) review the extent to which such energy consumption is considered in energy-economy models.

We find that the mining industry is currently responsible for approximately 1.7% of global final energy consumption. However, the mining industry's final energy consumption is likely to increase significantly, by a factor in the range 2–8 by 2060, depending on the future economic trajectory, on the evolution of energy intensities, and on future recycling rates. We also find that mineral material flows and their associated energy requirements (including the mining industry's energy consumption) are insufficiently covered in many energy-economy models. Our work suggests that the limited representation of material flows and associated energy requirements is currently an important blind spot in energy-economy modelling and may hinder the efforts of the community to build consistent energy transition pathways.

#### 1. Introduction

#### 1.1. Mining, energy consumption, and environmental impacts

The mining industry is responsible for a wide range of environmental and societal impacts, including biodiversity loss (Sonter et al., 2020; Luckeneder et al., 2021), soil and water pollution (Rasafi et al., 2021; Bardi, 2014), water consumption (Mudd, 2010; Aitken et al., 2016), and health impacts in workers and neighbouring communities (Stephens et al., 2001; Entwistle et al., 2019). In particular, the mining industry consumes considerable amounts of energy, which results in significant greenhouse gas emissions (Norgate et al., 2007; Norgate et al., 2011; Rankin, 2011). The magnitude of all these impacts has considerably increased over time as the quantities of extracted minerals has surged since the industrial revolution (Krausmann et al., 2009; Krausmann et al., 2018), driven by industrialisation and rapid economic growth (Steinberger et al., 2013; Wiedmann et al., 2015). The large environmental impacts caused by mining activities are of particular concern in the present context of major environmental degradation and transgression of some planetary boundaries that underpin the state of the Earth System (Steffen et al., 2015; Steffen et al., 2015; Ceballos et al., 2017), and in particular, climate change. The Paris Agreement was reached with the aim of limiting global warming to well below 2°C (compared to pre-industrial levels), and at pursuing efforts to limit the Earth's warming to 1.5°C (Rogelj et al., 2016; Schleussner et al., 2016). Achieving such ambitious climate targets requires a steep reduction in

https://doi.org/10.1016/j.gloenvcha.2023.102745

Received 27 January 2022; Received in revised form 21 July 2023; Accepted 6 September 2023 Available online 16 November 2023

0959-3780/© 2023 The Author(s). Published by Elsevier Ltd. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

<sup>\*</sup> Corresponding author. *E-mail address:* e.aramendia@leeds.ac.uk (E. Aramendia).

greenhouse gases until net-zero emissions are reached (Masson-Delmotte et al., 2021), which, in turn, requires a rapid curtailment in fossil fuel consumption (Smith et al., 2019). Limiting the level of global energy consumption is needed to help abate fossil fuel consumption, and hence, achieve net-zero targets.<sup>1</sup> Understanding the future possible energy pathways of energy-intensive industries, including the mining industry, is therefore key to supporting climate change mitigation efforts.

Despite the need for climate change mitigation, current trends suggest a future increase in the scale of mining activities (measured by extracted volumes - note that fossil fuel extraction activities are outside the scope of this study) and associated energy consumption, due to three main drivers. First is future economic growth, which has been shown historically to be highly correlated to energy and resources use (Stern and Kander, 2012; Hickel et al., 2019) - and thus resources extraction and mining activities. As the evidence of absolute decoupling between natural resources use and economic growth remains, at a global level, very thin (Bithas et al., 2018; Parrique et al., 2019; Haberl et al., 2020), future economic growth is likely to increase the scale of global mineral extraction. In addition, the growth in the information and communications technology sector, which has high demand for many minerals and metals (Gadgets et al., 2015), is likely to hamper such decoupling. Second is the renewable energy transition. Renewable energy systems are highly dependent on critical raw minerals (Kleijn et al., 2011; Moss et al., 2013; Valero et al., 2018), and their deployment will induce a surge in the demand and extraction of particular materials (Viebahn, 2015; de Koning et al., 2018; Elshkaki et al., 2019; IEA, 2021), potentially leading to bottlenecks in the supply chain and availability issues for specific materials (Valero et al., 2018; Watari et al., 2018; Capellán-Pérez et al., 2019; Moreau et al., 2019). Third comes the increase in the final energy required per unit mass of mineral recovered - i.e. the energy intensities of mining activities - associated with the decrease in natural resources deposits qualities. As high quality deposits tend to be exploited first, sustaining mineral extraction requires a move towards lower quality deposits (lower ore grades, lower grinding size, deeper and increasingly remote mines...), which in turn augments the energy consumption (as well as other environmental and social impacts) of mineral extraction (Cook, 1976; Prior et al., 2012; Bardi, 2014). Such a situation, characteristic of the ongoing process of natural resources depletion (Bardi, 2013; Pigneur, 2019), is likely to happen to numerous minerals as the quality of tapped deposits decreases (Mudd, 2010; Mudd, 2007; Mudd, 2010; Northey et al., 2014; Calvo et al., 2016), and is already limiting the productivity growth of the mining industry (Topp, 2008).

The three drivers of increasing mining activities and associated final energy use can conceptually be summarised by Eq. 1:

$$E = E_{growth} + E_{transition}$$

$$= \sum_{m} GDPi_{m}(1 - r_{m})e_{m} + \sum_{l,m} c_{l}j_{l,m}(1 - r_{m})e_{m}$$
(1)

where E stands for the mining industry's final energy consumption,  $i_m$  for the material intensity of the economy for mineral m,  $r_m$  for the recycled content (or recycling input rate) of mineral m,  $e_m$  for the final energy intensity of mining mineral m,  $c_t$  for the newly installed technology t, and  $j_{t,m}$  for the material intensity in mineral m of technology t. Only recycling rates appear as a potential offsetting lever. Taken together, these three drivers suggest that the energy consumption of the mining industry is likely to increase. In this context, it is crucial to

explore the future pathways that the mining industry's energy consumption may follow.

#### 1.2. Aim, approach, and content

The aims of the paper are threefold. First, to estimate the historical global (1971-2015) final energy consumption of the mining industry. Second, to explore the range of future possible pathways for the mining industry's final energy consumption to 2060. These pathways are based on a range of socio-economic scenarios taken from the Integrated Assessment Models (IAMs) literature, combined with different assumptions regarding the recycling rates of minerals and the increase in the final energy intensities of mining activities (denoted as energy intensities in the rest of the article). Third, to explore the extent to which the mining industry's global energy consumption and the energy requirements of material flows are incorporated in a number of influential energy-economy models, and discuss their treatment in the light of the future pathways previously constructed. The paper is structured as follows. Section 2 reviews the literature related to energy consumption of mining activities. Section 3 presents the methodology used for the historical and prospective analysis of the mining industry's final energy consumption, and Section 4 presents the results (first and second aims). Section 5 reviews the consideration of the mining industry's energy requirements in energy-economy models, as well as the consideration of broader material flows and associated energy requirements (third aim). Then, Section 6 discusses our findings and Section 7 presents the conclusions of the study. Appendix A presents a short definition of the energy-related terminology used throughout the paper.

#### 2. Mining and associated energy use: a review

Although the process of mineral extraction differs between minerals and extraction techniques, it may be decomposed in the following steps: (i) ore extraction, (ii) ore beneficiation, or concentration, and (iii) concentrate refining, which are represented in Fig. 1, and further described in Appendix B. The first two stages generally occur at the mine location and are considered part of the mining process, while the last stage usually occurs in a downstream metallurgical plants. In this article, we define the mining industry as the ore extraction and ore beneficiation stages for all minerals, excluding fossil fuels.<sup>2</sup>

#### 2.1. The energy consumption of the mining industry

#### 2.1.1. Global estimates

Global estimates of the energy consumption of the mining industry are scarce. The International Energy Agency (IEA), in its yearly World Energy Extended Balances (WEEB) (IEA, 2019) reports such energy consumption to be slightly below 1% of total final energy consumption. Such an estimate is likely to be somewhat underestimated, as mining activities are occasionally informal (Lahiri-Dutt, 2018), meaning that their energy consumption may not be reported by national statistical agencies to the IEA.<sup>3</sup> Another important note on the IEA dataset is that the accounting method only includes the *direct* energy used (energy used in situ by an industry or activity – see Appendix A for a definition of the energy-related terminology used throughout the paper), and excludes the *indirect* energy used (energy used by an industry or activity's supply chain), which avoids double accounting of energy flows in different enduse sectors. Next, some studies have assessed the global energy consumption of primary metal production using a whole supply chain

<sup>&</sup>lt;sup>1</sup> Of the four marker scenarios used in the Intergovernmental Panel on Climate Change's Special Report on 1.5°C, the two that do not rely on speculative carbon dioxide removal technologies assume a decrease in final energy consumption [(Masson-Delmotte et al., 2018) Chap. 2]. (See (Anderson and Peters, 2016; van Vuuren et al., 2017; Vaughan and Gough, 2016; Heck et al., 2018; Smith et al., 2016) for a critical discussion on carbon dioxide removal technologies.)

<sup>&</sup>lt;sup>2</sup> Categories 07, 08 excluding 0892, and 099 of the United Nations International Standard Industrial Classification (United Nations, 2008).

 $<sup>^3</sup>$  Note that the methodology we describe in Section 3 does not fully address this issue, which is due to the availability of data. We further discuss this in Appendix E.



Fig. 1. Graphical representation of the mineral extraction process. The mining activities cover the ore extraction and ore beneficiation steps, while the refining process, mostly relevant for metals, belongs to the metallurgical industry. Energy intensities only refer to the ore extraction and ore beneficiation stages in this paper.

perspective (accounting for both direct and indirect energy requirements). For instance, Bardi (2013) estimates the energy consumption of metal production around 5-10% of Total Primary Energy Supply (TPES), and Bihouix and De Guillebon (2012) around 8-10%. Similar values are obtained by Nuss et al. (2014) for the whole metal production process. We note, however, that these studies fail to differentiate such energy consumption in terms of mining activities, and downstream metallurgical processes, which explains the very large difference to the much lower values reported by the IEA (the scope adopted by these studies, which include indirect energy, as well as the quantification in terms of primary energy, also contribute to the difference, but to a lesser extent). Holmberg et al. (2017) attempted to fill this gap with a top-down approach, and estimated the energy consumption of global mining activities to be around 12 EJ, i.e. 3-4% of total final energy consumption, although we note that the adopted approach relies on the IEA's WEEB.

#### 2.1.2. Determinants of energy consumption of mining processes

Different factors determine the energy consumption of the mining industry. First, ore grades (i.e. the share, in mass, of valuable mineral contained in the mined ore) are inversely correlated to energy consumption (Vidal et al., 2021) - see Page et al. (1975) and Cook, 1976 for early studies. Such a relationship has been shown empirically in the case of copper (Norgate et al., 2010; Norgate et al., 2010; Northey, 2013; Calvo et al., 2016; Koppelaar and Koppelaar, 2016), gold (Muller et al., 2010; Norgate et al., 2012; Calvo et al., 2016), nickel (Norgate et al., 2010; Norgate et al., 2011; Norgate et al., 2011), zinc (Norgate et al., 2011; Calvo et al., 2016; Mudd et al., 2017), lead (Norgate et al., 2011; Mudd et al., 2017), rare earth elements (Weng et al., 2016), platinum group metals (Glaister et al., 2010), and uranium (Norgate et al., 2014; Parker et al., 2016). Indeed, as ore grades decrease, more ore needs to be extracted, moved, hauled and crushed, to obtain the same amount of valuable mineral. Second, the grind size of the extracted ores, which determines the size to which mineral particles must be ground to separate valuable minerals from the gangue (rest of the mined ore), is a key parameter (Norgate et al., 2010; Norgate et al., 2011; Haque et al., 2015). Indeed, the smaller the particles need to be ground to liberate the element of interest, the higher the required energy consumption in the comminution (reduction in particles sizes) process (Morrell, 2004). Third, the mine depth is also an important determinant; the deeper the ore needs to be extracted from, the higher the energy consumption of the mining process. Koppelaar and Koppelaar (2016) quantify the "interactive effect between mine depth and ore grades", and finds significant influence of mine depth. Fourth, the energy consumption of mining activities depends on the extraction technique employed. Norgate et al.

(2010) state that "the most appropriate route [...] in terms of embodied energy and greenhouse gas emissions depends on the mineralogy of the ore deposit concerned." Hence, the technique that minimises energy consumption depends on the type of ore, ore grade, grind size, recoverable by-products, and other characteristics of the ore deposit (Norgate et al., 2011; Koppelaar and Koppelaar, 2016). Lastly, the effects of technological improvements and innovation are also important, as new extraction techniques and improvements in the machinery used may result in increases in energy efficiency and hence, in lower energy intensities (US-DoE, 2007; Rötzer et al., 2020; Vidal et al., 2021).

#### 2.2. Increase in energy intensities of mining activities

#### 2.2.1. Historical evolution of energy intensities

Although there is evidence that average ore grades have decreased over time in the case of numerous minerals, there are only rare studies exploring the historical evolution of average energy intensities with a national or global scope. First, Calvo et al. (2016) finds an increase of 12% in the Chilean copper energy intensities in the period 2003-2013 using company and mine level data (average yearly increase of 1.1%). Second, the Chilean Copper Commission (2020) reports national copper production and energy use over time; these data show that the energy intensity of copper mining is found to increase by 66% in the period 2001-2019 (average yearly increase of 3.0%; the historical evolution can be found in Appendix C). Third, Rötzer et al. (2020) adopt a processbased approach to quantify the energy intensities in the copper mining industry in the 1930s, 1970s, and 2010s. The authors show how the relationship between ore grade and energy intensity changes over time with the effect of technological improvements, and find that technological improvements have considerably reduced energy intensities of copper mining despite decreasing ore grades between the 1930s and 1970s. Conversely, the scale of the effects of technological improvements has been much more limited between the 1970s and 2010s, leading to an increase of approximatively 30% in average copper energy intensities. Hence, studies point to an increase of energy intensities in recent years, meaning that the influence of geological factors and depletion has outstripped technological improvements. Such a situation may well carry on, as there are thermodynamic and practical limits to the energy efficiency of processes (US-DoE, 2007; Paoli et al., 2020; Vidal et al., 2021).

#### 2.2.2. Forecasting future evolutions of energy intensities

The few studies that have attempted to model future increases in energy intensities of mining activities proceed in two steps: (i) extrapolation of the future ore grade of a given mineral, and (ii) determination of the energy requirements based on the ore grade-energy relationship for that given mineral.<sup>4</sup> The evolution of future ore grades may be determined in two ways. First are studies that use a time-dependent function fitted to historical data, but independent of future cumulative extraction. van der Voet et al. (2019) modelled first ore grades as a decaying exponential with respect of time to assess environmental impacts of major metal production. A similar method was used later by Kuipers et al. (2018) and Dong et al. (2020), who assessed the environmental impacts of future copper production globally and in China, respectively. Second are studies that adopt a mechanistic approach, linking cumulative extraction of a given mineral to the ore grade, hence assuming an ore grade-tonnage distribution for the mineral under study. Harmsen et al. (2013) explicitly construct a global ore grade-tonnage distribution for copper, and assess how future extraction will increase copper energy intensities. Elshkaki et al. (2016) and Ciacci et al. (2020) both link the average copper ore grade to cumulative copper production in order to estimate environmental impacts of copper production globally and in the EU-28, respectively.

There are however large uncertainties associated with the estimation of the future evolution of energy intensities. First, because there are only few studies and data available regarding the historical evolution of energy intensities, and because the increases in energy intensities affect differently each mineral, mine and country, which complicates the extrapolation of particular case studies to the broader mining industry. Second, modelling the future ore grade for each mineral remains a complex task. To proceed to such a forecast, one needs to either model the ore grade as a function of time, or to link the cumulative extraction to the ore grade through an ore grade-tonnage distribution. While the second approach is endogenous and more accurate than the time dependency function, it comes with significant uncertainties associated with the construction of ore grade-tonnage distributions. Whether ore grade-tonnage distributions follow a unimodal (Lasky, 1950) or bimodal (Skinner, 1976) distribution remains a matter of scientific debate (Arndt et al., 2017), and may well have dramatic implications for the future evolution of energy intensities - see Appendix D. Hence, there is large uncertainty associated with the future evolution of ore grades, which depend on unknown geological factors.

#### 3. Methodology

This section introduces the methodology followed in this study. The data used and data sources are described in Appendix E. We cover 59 minerals, the full list can be found in Table F.1.

#### 3.1. Historical final energy consumption of the mining industry

The historical final energy consumption of the mining industry (in Joules) for a mineral *m* is simply determined as:

$$E_{m,t} = P_{m,t} f_m, \tag{2}$$

where  $P_{m,t}$  (kilograms) is the primary extraction of mineral *m* in the year *t*, and  $f_m$  (Joule/kilogram) is the historical (and constant over time) final energy intensity (in the rest of the paper, energy intensity refers to *final* energy intensity, unless stated otherwise) of mineral *m*. We use energy intensities independent of time for the historical analysis because (i) uncertainties in the estimates were too significant to produce robust time series for energy intensities, and (ii) uncertainties related to the historical values of energy intensities are covered by the sensitivity analyses (using a Monte Carlo simulation) introduced subsequently.

Historical energy intensities by extracted mineral. The energy

requirements of mining activities may be distinguished in terms of *direct* and indirect energy requirements [(Rankin, 2011, Chap. 9)], and are usually quantified in terms of primary energy requirements using Life Cycle Analysis methods. Direct energy requirements refer to the energy used in situ to operate the mine. Conversely, indirect energy requirements refer to the energy used in the mine supply chain, but ex-situ, to provide inputs needed to operate the mine (e.g. chemicals, machinery...). When adding direct and indirect energy requirements (quantified as primary energy requirements), the Gross Energy Requirement (GER) is obtained.<sup>5</sup> As both the direct and indirect (due to increasing requirements of non-energy inputs as well) energy requirements of mining are likely to increase as a result of mineral depletion, we adopt the GER as a measure of *primary* energy intensity of each mineral.<sup>6</sup> We review the literature to estimate the GER of each mineral using the studies of Hammond et al. (2011); Norgate et al. (2011); Rankin (2011); Mudd (2012); Norgate et al. (2012); Nuss et al. (2014); Calvo et al. (2018). Then, we determine the average final-to-primary energy ratio for the global mining and quarrying sector for the period 2000-2015 (the ratio is very stable in that time period, with an average value of 0.58) using a recently developed Physical Supply Use Table framework (Heun et al., 2018; Aramendia et al., 2022) (an explanation of the method and calculations is available in SI-1), and we multiply the GER by this average ratio to determine the historical *final* energy intensity for each mineral. The values used as GERs, as well as details on the estimation of each GER are available in SI-2.

Sensitivity analysis regarding historical energy intensities. We analyse the sensitivity of results to the historical energy intensity  $f_m$  chosen for each mineral m. We do so by conducting a Monte Carlo simulation (1000 runs) where each mineral's historical energy intensity  $f_m$  follows a normal probability distribution function – see SI-1 for more information on the standard deviations used.

#### 3.2. Estimating the future final energy consumption of mining activities

Fig. 2 summarises the four step approach for generating future pathways for the mining industry's final energy consumption. First, we determine future mineral demand for the rest of the economy using six different socio-economic scenarios, which provide global GDP and final energy projections. Second, we determine future mineral demand for Energy Transition Technologies (ETTs)<sup>7</sup> using a high and low bound of mineral demand for the development of ETTs, to capture the uncertainty related to the mineral requirements of the energy transition (see for instance Beylot et al. (2019) for a quantification of such uncertainty). Third, we determine the required primary mineral extraction using three recycling rates scenarios (constant, moderate increase, and high increase). Fourth, we determine the mining industry's energy requirements, using three energy intensities scenario (constant, low increase, and high increase). Hence, we obtain 54 possible combinations of socio-economic, recycling rates, and energy intensities scenarios, each combination having a high and low bound of mineral demand for ETTs (hence 108 total pathways).

Socio-economic scenario selection. To explore future pathways, we

<sup>&</sup>lt;sup>4</sup> We also note the work of Fizaine et al. (2015), who use a different approach, as they assess the sensitivity of renewable energy systems energy returns as function of the average decline in ore grades, independently of time.

<sup>&</sup>lt;sup>5</sup> Note that the GER indicator may be called the Cumulative Energy Demand (CED) in the Life Cycle Analysis terminology.

<sup>&</sup>lt;sup>6</sup> The estimation often involved separating the primary energy requirements of the mining and metallurgical steps (for metal production) using a figure showing the breakdown in Nuss et al. (2014).

<sup>&</sup>lt;sup>7</sup> Material requirements are considered for the following technologies according to the review of Watari (2020): solar photovoltaic, concentrated solar power, wind onshore and offshore turbines, electric vehicles, fuel cells, nuclear, geothermal, biomass, CCS, as well as additional grids and storage batteries needed for the deployment of renewable energy. Note that the scope adopted does not cover all mineral demand required for a low carbon economy, for instance mineral demand for the insulation of houses or for heat pumps is not covered.



Fig. 2. Workflow for constructing future pathways of the mining industry's global final energy consumption, and possible combinations of socioeconomic scenario, recycling rates scenario, mining energy intensities scenario, and mineral requirements for ETTs. ETTs: Energy Transition Technologies.

select the Beyond 2 Degrees Scenario (B2DS) developed by the IEA (IEA, 2017), as well as the four 1.5°C consistent marker scenarios chosen by the IPCC in the Special Report on 1.5°C [(Masson-Delmotte et al., 2018) Chap. 2], three of which are based on the Shared Socioeconomic Pathways (SSPs) (O'Neill et al., 2017); SSP1 (Fujimori et al., 2017), SSP2 (Fricko et al., 2017), SSP5 (Kriegler et al., 2017), and the last being the Low Energy Demand scenario (Grubler et al., 2018). In addition, we recognise the validity of the recent argument in favour of exploring a broader set of socio-economic scenarios and of exploring post-growth scenarios (Keyßer and Lorenz, 2021; Hickel et al., 2021). Hence, we also define and include a post-growth socioeconomic scenario, in which global GDP declines by 0.2% yearly.

#### 3.2.1. Estimation of future mineral demand for the rest of the economy

Considering the strong historical link between economic activity and material consumption (Krausmann et al., 2018), we follow the approach developed in Capellán-Pérez et al. (2019) to estimate mineral demand for the rest of the economy by constructing for each mineral a linear model of mineral demand as function of economic activity, accounted for by the GDP metric. Hence, the demand for mineral m (in kilograms) for the rest of the economy over time is determined as:

$$D_{m,RoE} = a_m y + b_m, \tag{3}$$

where *y* stands for GDP (constant 2010 \$US), and  $a_m$  and  $b_m$  are determined for each mineral *m* fitting historical data.<sup>8</sup> For the few minerals that do not correlate well to GDP – in general the linear regression works remarkably well – we use a constant demand equal to the average demand in the period 2005–2015. The values of the linear regression coefficients, the  $r^2$  coefficient, the historical period for fitting data, and the

#### Table 1

Sources for mineral demand by the deployment of Energy Transition Technologies, for the high and low bound of mineral demand. The future demand of minerals that are covered in the review conducted in Watari (2020) follow the pattern described in column (a), and the future demand of minerals that are not covered in the review but that are covered in the work conducted in Capellán-Pérez et al. (2019) follow the pattern described in column (b).

Mineral demand	(a) Review by Watari (2020)	(b) Study by Capellán-Pérez et al. (2019)
Low	Lowest value of the studies reviewed	Demand in the scenario with transition to a 50% renewables energy mix in 2060.
High	Highest value of the studies reviewed	Demand in the scenario with transition to a 100% renewables energy mix in 2060.

demand value used for minerals for which demand is kept constant, are reported in Appendix F.

#### 3.2.2. Estimation of future mineral demand for ETTs

To estimate the future mineral demand for the energy transition, we use first the literature review of the critical metal requirements for the energy transition conducted in Watari (2020), which summarises the metal requirements for the energy transition to 2050 determined by numerous studies, for a range of metals. Second, for those minerals that are not covered in the review, we use the study conducted in Capellán-Pérez et al. (2019), which estimates the mineral requirements of the energy transition for three scenarios (respectively 50%, 75%, and 100% renewables in 2060). Then, we define both a high and a low future mineral demand for the ETTs following Table 1 – see SI-1 for more information.

#### 3.2.3. Estimation of future primary mineral extraction

The future primary mineral extraction of mineral *m* (in kilograms) is then determined as:

<sup>&</sup>lt;sup>8</sup> We note that rigorously, one would need to subtract the GDP due to the development of the ETTs, the material requirements of which are accounted for according to Section 3.2.2.

#### Table 2

Description of the evolution of recycling rates (recycled content) over the period 2015–2060 for the three recycling rates scenarios. The limit on recycled content does not apply to initial recycled contents because all are below 80%.

Recycling rate scenario	Recycled content evolution	Limits on recycled content	Comment
Constant	Recycled content remains equal to its initial value.	No limit.	Pessimistic scenario
Moderate increase	Initial recycled content increases by 50%.	If recycled content reaches 80%, it is then held constant.	Realistic scenario
High increase	Initial recycled content doubles. If the doubling of the initial recycled content does not reach 30%, the recycled content is set to increase linearly until reaching 30% in 2060.	If recycled content reaches 80%, it is then held constant.	Optimistic scenario

$$P_{m,t} = D_{m,t}(1 - r_{m,t}), \tag{4}$$

where  $D_{m,t}$  (kilograms) represents the total demand for mineral *m* at *t*, and  $r_{m,t}$  (without unit) represents the recycled content of mineral *m* at *t*, i.e. the share of recycled material *m* relative to total new consumption of material *m*. Next, we define the three recycling rates scenarios presented in Table 2: the constant recycling rate scenario, where no improvements are made, the moderate increase scenario, where recycling rates (modelled as recycled content) increase by 50% in the period 2015–2060, and the high increase scenario, where recycling rates (modelled as recycled content) double in the period 2015–2060 – initial recycling rates are taken from Graedel et al. (2011) and Haas et al. (2005), and can be found in SI-2.<sup>9</sup> We set a maximum limit of 80% recycled content for all minerals to account for the fact that some applications require very high purity materials, which can only be supplied by primary production.

*3.2.4. Estimation of the future mining industry's final energy consumption* We apply Eq. 5 to determine the future final energy consumption of mining activities (in Joules):

$$E_{m,t} = P_{m,t} e_{m,t},\tag{5}$$

where  $e_{m,t}$  (Joule/kilogram) stands for the future *final* energy intensity of mining mineral *m*, at a given time *t*, defined thereafter.

*Future energy intensities.* The energy intensity  $e_{m,t}$  associated to mineral *m* at a given time *t* is modelled to vary over time as:

$$e_{m,t} = f_m \alpha_{m,t},\tag{6}$$

where  $f_m$  (Joule/kilogram) is the historical energy intensity of extraction of mineral m, i.e. the likely future energy intensity of mineral m if depletion effects were not at play, and  $a_{m,t}$  a dimensionless coefficient modelling the increase in the energy intensity of mineral m in year t.<sup>10</sup> To model the increase in energy intensities, we begin by classifying minerals in terms of minerals that are likely to be affected by decreasing ore deposit qualities (e.g. copper, silver, zinc, etc.), and those that are not likely to be affected by decreasing ore deposits (e.g. sand, gravel, limestone, etc.). The classification is done using various information found in the literature and expert judgment, and is fully described in SI-2. To deal with the uncertainty associated with increasing future energy intensities (see Section 2.2.2), we define three different scenarios. In the

"constant intensities" scenario, which should be interpreted as a baseline scenario assuming the absence of mineral depletion effects (or their full compensation by technological improvements), energy intensities are held constant over time. In the "low increase" scenario, energy intensities start increasing around 2000 and follow a linear trend fitted to the 12% increase reported in Calvo et al. (2016) for the period 2003-2013. In the "high increase" scenario, energy intensities start increasing around 2000 and follow a linear trend fitted to the 66% increase determined from the Chilean Copper Commission in the period 2001-2019 (Chilean Copper Commission, 2020). Fig. 3 shows the evolution of  $\alpha_{m,t}$ , for those minerals affected by depletion, in each of the three scenarios. We also note that the  $\alpha_{m,t}$  coefficient, as it is constructed from empirical historical data represents the combined effects of mineral depletion and of technological improvements, which are also accounted for. Appendix C shows that our increasing energy intensity scenarios are in line (in terms of magnitude) with other studies.

#### 3.3. Methodology limitations

#### 3.3.1. Exogenous inputs

Our methodology uses several exogenous parameters (recycling rates, future energy intensities, mineral demand for ETTs), while these parameters are endogenous parameters and dependent on the socioeconomic scenario. As this article aims at exploring the range of possible future pathways for the mining industry's final energy consumption, and not at accurately modelling the future, we believe that such assumptions are appropriate. However the following points should be noted:

- *Exogenous future recycling rates*. Recycled minerals are in reality a function of both end-of-life recycling rates and minerals reaching their end-of-life from in-use stocks, which represents a limiting factor to the maximum amount of minerals that can effectively be recycled. Hence, some of the recycling rates scenarios (especially, the high recycling rate scenario) may not be realistic, particularly when combined with high mineral demand (i.e. high economic growth) socio-economic scenarios, because there may not be enough minerals reaching their end-of-life from in-use stocks to reach such recycled contents.
- *Exogenous future energy intensities*. Future energy intensities account for the fact that extracting minerals from deposits becomes harder as a result of the depletion of higher quality deposits. Hence, the higher cumulative extraction is, the faster will energy intensities increase. Consequently, energy intensities are in reality function of future mineral demand, of the amount of minerals that can be supplied through recycling processes, and of geological factors. In addition, the constant energy intensities that we use for fairly abundant mineral assume that technological improvements will be sufficient to offset the effects of mineral depletion, but such energy intensities may even decrease if technological improvements outstrip depletion effects.
- Exogenous future mineral demand for ETTs. In a similar vein, the mineral demand for developing ETTs is a function of the socioeconomic scenario, particularly in terms of final energy demand and of the ETTs that are deployed to fulfil such final energy demand.

#### 3.3.2. Mineral demand modelling

The approach we follow to determine mineral demand for the rest of the economy (Eq. 3) does not explicitly represent future trends, such as digitalisation, increasing demand for specific metals like Rare Earth Elements – such trends are only partly captured by our modelling of mineral demand. Likewise, our approach does not take into consideration structural changes that may allow economic growth to be at least partly decoupled from mineral demand in the future, such as in-use stocks saturation (Pauliuk et al., 2013; Bleischwitz et al., 2018), material demand saturation (Ruijven et al., 2016), structural economic

<sup>&</sup>lt;sup>9</sup> The dynamic recycling rates are thus mineral specific, and evolve by a given yearly percentage based on the initial recycling rate.

<sup>&</sup>lt;sup>10</sup> We note that the adopted approach supposes that all energy inputs to the mining processes, be they direct energy in the form of fuel or electricity, or indirect energy embodied in other inputs or equipment, increase equally as a result of mineral depletion.



**Fig. 3.** Values of  $a_{m,t}$  for minerals affected by the decrease in mineral deposit qualities for the three increasing energy intensities scenarios. Constant scenario: energy intensities are kept constant for all minerals. Low increase: energy intensities start increasing around the year 2000 and are extrapolated following the trend reported in Calvo et al. (2016). High increase: energy intensities start increasing around 2000 and are extrapolated following the trend calculated from the Chilean Copper Commission (2020) data.

changes towards less material intensive sectors (Pothen, 2017), or increases in the material efficiency with which services are delivered (Allwood et al., 2013; Carmona et al., 2017). Particularly, this implies that our approach for mineral demand modelling may not be appropriate for those socioeconomic scenarios that inherently assume a high material-GDP decoupling, such as the LED or B2DS scenarios. However, while such decoupling may indeed occur to some extent, we argue that unless the underlying dynamics are made explicit in energy-economy models, one can legitimately think that historical trends will continue. Extrapolating historical trends is thus appropriate to determine the importance of explicitly modelling the mining industry's final energy consumption.

#### 4. Results

#### 4.1. Historical final energy consumption of the mining industry

Fig. 4.a shows the historical final energy consumption of the mining industry obtained with constant energy intensities alongside the 90% and 95% confidence intervals obtained with the Monte Carlo simulation. The final energy consumption of the mining industry has increased considerably in the period 1970-2015, and has reached about 6.6 EJ in 2015. The uncertainty associated with such an estimate is however substantial, and the Monte Carlo simulation yields as 95% confidence interval a range of 5.2-8.0 EJ. Then, Fig. 4.b shows the share of global final energy consumed by the mining industry. Such share has increased from around 1.1% in 1970 to around 1.7% in 2015. Likewise, we quantify as the 95% confidence interval a range of 1.3-2.0% of global final energy consumption in 2015. The final energy consumption of the mining industry is therefore a small share of the global final energy consumption, although higher than the value (0.75%) reported by the IEA (see Section 2.1.1 for the limitations of IEA data), as shown in Table 3. As explained in Section 2.1.1, a key reason is the fact that the IEA's methodology only accounts for direct energy use (to avoid double accounting across end-use sectors), while our calculations include indirect energy, i.e. final energy used by other industries that are part of the mining industry's supply chain.

Fig. 5 shows the breakdown of final energy consumption by mineral<sup>11</sup> – note that the breakdown by mineral should be interpreted with caution due to the uncertainty of the historical energy intensities  $f_m$ estimated for each mineral. The extraction of just a few minerals (aluminium, clays, copper, gold, iron ore, limestone, Platinum Group Metals (PGMs), sand and gravel, and silver) appears to be responsible for around 90% of the mining industry's final energy consumption. Construction minerals (clays, limestone, sand and gravel) are responsible for a large share of final energy consumption despite low energy intensities, because of the large tonnages extracted yearly. Precious metals (gold, silver, and PGMs) are also responsible for a large share of final energy consumption, but for the opposite reasons: although they are extracted in small amounts, they require large amounts of energy to be extracted. Then, ferro-alloy metals and non-ferrous metals each contribute to a considerable share of final energy consumption, which is mostly due to the extraction of iron ore, chromium, molybdenum and nickel in the case of ferro-alloy metals, and of aluminium, copper, and zinc in the case of non-ferrous metals.

#### 4.2. Future pathways for the mining industry's final energy consumption

Fig. 6 summarises the increase of the mining industry's final energy consumption in 2040 and 2060 relative to 2015, for all constructed pathways (54 scenario combinations, each with a high and low bound of mineral demand for ETTs, so 108 total pathways). The mining industry's final energy consumption increases significantly in almost all cases: by a factor in the range 2–6 for high growth scenarios (i.e. SSP2, LED, B2DS, SSP1), and by a factor in the range 3–8 in the case of the extremely high growth SSP5 scenario. Only in the case of the post-growth scenario are there moderate changes in the mining industry's final energy consumption, reaching at most a 50% increase, and remaining constant or decreasing in some cases.

Fig. 7 shows the influence of each parameter by displaying (a) the cumulative final energy consumption and (b) the final energy consumption in 2060 in the case of the B2DS socio-economic scenario

 $<sup>^{11}</sup>$  See SI-1 for the evolution of the breakdown over time and for the breakdown by mineral group.



Fig. 4. Final energy consumption of the mining industry (left), and share of global final energy consumption consumed by the mining industry (right). Note that calculated values account for both direct and indirect energy, while the IEA's accounting method only includes direct energy.

#### Table 3

Historical final energy consumption of the mining industry alongside values reported by the International Energy Agency. Both default and Monte Carlo analysis results are shown. Note that calculated values account for both direct and indirect energy, while the IEA's accounting method only includes direct energy. Values in EJ (J.1e18).

Value	1971	1980	1990	2000	2010	2015
Default estimation	1.93	2.39	3.01	3.86	5.59	6.64
Low bound, 90% confidence	1.58	1.97	2.47	3.12	4.54	5.42
High bound, 90% confidence	2.27	2.82	3.52	4.55	6.60	7.81
Low bound, 95% confidence	1.51	1.88	2.39	3.01	4.35	5.16
High bound, 95% confidence	2.34	2.90	3.62	4.68	6.78	8.05
IEA	0.625	1.08	1.64	1.64	2.55	3.03

(results for other socio-economic scenarios are available in SI-1), as function of the recycling rates and energy intensities scenarios (graph for the high bound value of mineral demand for ETTs). Results show that even when using the high bound of mineral demand due to the energy transition, the rest of the economy remains responsible for a very large majority of the mining industry's final energy consumption - the main driver for such energy consumption is global economic output. The influence of the future evolution of recycling rates and energy intensities is considerable, particularly when looking at the final energy consumption in 2060. The magnitude of increasing energy intensities affects the results considerably. The high increase in energy intensities scenario yields a final energy consumption in 2060 higher than the low increase and the constant energy intensity scenarios by respectively 35% and 60%. Increases in recycling rates can help to reduce the primary extraction and hence the mining industry's final energy consumption. Indeed, the moderate increase in recycling rates scenario reduces final energy consumption by approximatively 20% compared to the constant recycling rates scenario in 2060, and the high increase in recycling rates scenario, by 40%.

Fig. 8 shows the future energy pathways for each combination of

socio-economic, energy intensity, and recycling rates scenarios, when considering the high bound for mineral demand for the ETTs. (See SI-1 for low bound and for energy use breakdown by mineral.) Results show that the socioeconomic scenario is of critical importance - only the postgrowth scenario limits the increase in the mining industry's final energy consumption. Hence, the future economic trajectory, in terms of GDP, appears to be critical for the future pathways of the mining industry's final energy consumption with higher economic activity leading to higher mining industry final energy consumption. The figure shows how the influence of the future evolution of recycling rates and energy intensities considerably increases over time to very significant levels. The mining industry's final energy consumption increases slower from 2055 onwards (and even decreases in the post-growth scenario) due to the underlying ETTs development data (see Table 1), according to which most of the energy transition is accomplished by 2055 in the high range of mineral demand for the ETTs.<sup>12</sup> Table 4 then provides the maximum fraction of global final energy consumption devoted to the mining industry reached over the 2015-2060 time period, for each of the socioeconomic scenarios (see SI-1 for the evolution of the fraction over time in each scenario). Results reveal that in the high growth socioeconomic scenarios, if we consider that the low increase in energy intensities is at minima likely, the future mining industry's final energy consumption is likely to reach values in the range of 4-12% of forecasted global final energy consumption - depending on the trajectory of future recycling rates - and even higher in the most pessimistic scenarios (steep increase in energy intensities). Such values would be extremely high, and call into question whether the final energy projections reported by each socio-economic scenarios would then be followed - and hence

<sup>&</sup>lt;sup>12</sup> We note that such a trend is highly dependent on the pace of the energy transition, and there is no evidence that the final energy consumption due to the ETTs development will start to decline after 2055. However, the trend shows that once the energy transition is accomplished, the final energy consumption of the mining industry entailed by the ETTs will decrease.



Fig. 5. Breakdown of historical final energy consumption of the mining industry by mineral for the year 2015. PGMs: Platinum Group Metals.



**Fig. 6.** Final energy consumption of the mining industry in 2060 relative to 2015, for each different socio-economic, recycling rate, energy intensity, and renewable energy scenario. (2015=1.) Points with the same colour and shape correspond to the high and low range of mineral demand for the Energy Transition Technologies, with the same recycling rate and energy intensity scenarios. Energy intensity refers to the *a* variable.

whether climate targets would be reached – or whether global final energy consumption would be higher than forecasted. Conversely, in the case of a post-growth socio-economic development, the mining industry's final energy consumption is likely to remain below 4% of global final energy consumption, which is much closer to historical values.

#### 5. Mineral materials in energy-economy models: a review

#### 5.1. General approach

In many models, the mining industry is represented as an economic

sector, through its monetary output, which may be linked to the monetary output of other sectors (e.g. the construction sector). In this review, we focus on the extent to which the production of mineral materials (which includes their mining) is considered explicitly through physical quantities (i.e. in tonnes), because such an approach provides a more accurate representation of mineral material flows and associated energy requirements than using monetary values. The mining industry is hence considered in this review through the lens of the primary production of mineral materials, which is closely related to the consideration of broader material cycles in models, as the amount of extracted materials is a function of the mineral demand as well as of the amount of



**Fig. 7.** (a) Cumulative final energy consumption (2015–2060) and (b) Final energy consumption in 2060 for the Beyond 2 Degrees Scenario, as function of the recycling rates scenario, of the energy intensities scenario. Results for the high bound value of mineral demand for Energy Transition Technologies (ETTs). Energy intensity refers to the  $\alpha$  variable.

end-of-life materials that can be effectively recycled. Hence, we review the broader consideration of material cycles and associated energy requirements in models through the following four criterion: (i) the mineral materials covered, (ii) the description of material demand, (iii) the differentiation of primary mineral extraction and secondary production (recycling), and (iv) the feedback of material flows on energy consumption – Appendix G summarises the approach of each reviewed model in respect of each criterion.

#### 5.2. Review findings

We summarise here the findings of the review – implications for the modelling community are further discussed in Section 6.2.

*Mineral materials coverage.* Most models only consider cement and (or) steel. The Shell WEM, the E3ME model, and the GEM-E3/PRIMES model have a high mineral material coverage, although it comes at the expense of aggregating heterogeneous materials in broad categories (e.g. non-ferrous metals, industrial minerals, etc), which limits the precision with which material flows can be represented. The IEA's WEM

performs best in terms of mineral materials coverage and disaggregation, with steel, aluminium, copper, nickel, lithium, cobalt and rare earth elements covered for both the ETTs and the rest of the economy, and zinc, PGMs, manganese, graphite, and molybdenum are covered for the ETTs. The MEDEAS model also performs well in terms of mineral materials coverage and disaggregation, although it only translates material flows into energy requirements indirectly and partially, through the dynamic Energy Return On Investment of the energy system (Capellán-Pérez et al., 2019).

Consideration of mineral material stocks and flows. Only the IEA's WEM, and the IMAGE model explicitly consider the whole material cycles through stocks and flows, leading to a primary mineral material demand (and hence, feedback on energy consumption) consistent with mineral material stocks and flows, although only for a limited number of materials. In addition, work is under way in the GEM-E3/PRIMES model to describe material stocks and scrap availability, and in the IMAGE model to incorporate the stocks, flows and energy requirements of a larger set of mineral materials (see (Deetman et al., 2018; Deetman et al., 2020; Deetman et al., 2021)). Conversely, most models do not



Fig. 8. Future mining industry's final energy consumption pathways by socio-economic scenario, energy intensity scenario, and recycling rate scenario, when considering the high range for mineral demand for Energy Transition Technologies. Energy intensity refers to the  $\alpha$  variable.

differentiate between primary and secondary production, or do it using exogenous recycled content data or following historical trends.

Feedback of mineral material demand on energy consumption. There are three types of mechanisms through which mineral material demand is connected to energy consumption in the reviewed models. First are models (AIM/CGE and E3ME) for which mineral material demand increases the monetary sectoral output of relevant industries, and consequently, the energy requirements of such industries. Second are General Equilibrium Models (REMIND-MAgPIE and IMACLIM), for which an equilibrium between output and inputs (including energy requirements by energy carrier) is directly determined through an optimisation procedure, and hence for which the determination of the exact feedback of mineral material demand on energy requirements is complex. Third are the remaining models (excluding MEDEAS) which link mineral material demand to energy requirements through the use of energy intensities of production for each material (e.g. GJ per tons of primary or secondary steel produced) – such energy intensities may be broken down by energy carrier, may depend on the modelled technology mix or on exogenous assumptions, depending on the model. We note that no model considers the increasing energy intensities of mining activities.

#### Table 4

Maximum fraction of global final energy consumption reached by the mining industry's final energy consumption over the 2015–2060 time period. Values in percentages. Energy intensity refers to the  $\alpha$  variable.

Energy intensity scenario	Recycling rate scenario	PG	SSP2- 1.9	LED	B2DS	SSP1- 1.9	SSP5- 1.9
Constant	Constant Moderate High	2.95 2.44 1.86	5.51 4.62 3.57	10.78 8.98 6.89	7.26 6.07 4.70	9.11 7.55 5.79	7.12 5.89 4.52
Low increase	Constant Moderate High	3.43 2.84 2.18	6.37 5.31 4.12	12.66 10.42 7.98	8.38 6.99 5.42	10.73 8.83 6.69	8.36 6.88 5.21
High increase	Constant Moderate High	4.53 3.75 2.90	8.39 6.90 5.35	17.03 13.93 10.51	11.11 9.09 7.02	14.35 11.75 8.86	11.13 9.11 6.87

#### 6. Discussion

#### 6.1. Levers to limit the mining industry's final energy consumption

This study has shown that the mining industry's final energy consumption may increase considerably in the future, although there are large uncertainties associated with such a prospective analysis. However several levers can help limit such increases, which we critically discuss hereafter.

Innovation, technological improvements, and efficiency gains. Favouring innovation and energy efficiency in the mining sector may help to limit future increases in energy intensities, as there are still significant energy efficiency opportunities in the mining industry (US-DoE, 2007). Increasing the share of electricity as final energy carrier in mining activities may also contribute to limiting future increases in energy intensities, as electricity tends to be used with significantly higher efficiencies than fuels. However, current trends (see Section 2.2), at least for relatively scarce metals, indicate the predominance of geological factors over technical developments, thereby questioning the extent to which innovation and efficiency can limit future increases in energy intensities. For minerals affected by mineral depletion, energy intensities are likely to carry on increasing, (particularly as any technological improvements may be used precisely to mine lower quality deposits). Technological improvements may however be able to lower the energy intensities of fairly abundant minerals, although there are thermodynamic and practical minimum limits to energy intensities (US-DoE, 2007; Paoli et al., 2020; Vidal et al., 2021).

*Fostering recycling rates.* Fostering high recycling rates appears to be a key lever to reduce the mining industry's final energy consumption. Significantly increasing recycling rates obviously implies reaching high end-of-life collection rates, developing appropriate technologies and industrial sectors, but it also implies rethinking the current use of minerals, particularly in the case of metals used for high-tech applications. Indeed, some metals are consumed in multiple dispersive uses (Vidal et al., 2021), for which recycling is either altogether impossible, or is currently not achievable, and are hence "lost by design" (Ciacci et al., 2015).<sup>13</sup> Some other metals are used in extremely low concentrations, for instance in superalloys and high-tech applications, making their recycling very difficult. For some minerals the final use concentration may sometimes be lower than currently mined deposits concentrations,

so that the recycling process may even lose its energy saving and climate mitigation potential (Schäfer et al., 2020). Recycling may sometimes only be possible as nonfunctional recycling, whereby the mineral becomes an impurity and loses its functionality (for instance, minor metals alloyed to steel and recycled as secondary steel, mixed with different steel types). Reconsidering the extent to which these dispersive and hardly recyclable uses are employed appears crucial to reach high end-of-life recycling rates.

Future economic activity. In our analysis, global economic activity remains a chief determinant of mineral demand, and consequently, of the mining industry's final energy consumption. Only the post-growth socio-economic scenario limits the final energy consumption of the mining industry to a level comparable to its current value. In addition, it is worth noting that conversely to what is assumed in this study, energy intensity and recycling rates scenarios are not independent from the socio-economic pathway. Indeed, the higher mineral extraction is (e.g. because of a high economic growth), the quicker will ore quality deposits decrease, and hence the faster will energy intensities increase. Similarly, the higher mineral demand is, the lower the fraction of the demand covered by recycled minerals can be, and the lower the recycled content of consumed mineral materials can be. Hence, our results support the argument that exploring post-growth socio-economic scenarios as an approach to limiting environmental damage is an essential research direction for the energy-economy modelling community (Keyßer and Lorenz, 2021; Hickel et al., 2021).

# 6.2. The need for a consistent modelling of mineral material flows and associated energy requirements in energy-economy models

We have shown in Section 4 that the mining industry's energy requirements are likely to increase considerably in the future, and may reach very high levels if historical demand trends carry on. Then, Section 5 has shown that mineral material flows and their associated energy requirements, including the mining industry's energy requirements, are only described to a limited extent in the energy-economy models we reviewed. Hence, this paper has shown the need to move towards a more explicit and comprehensive consideration of material flows and of their associated energy requirements. Here, we suggest and discuss four principles for an improved modelling of material flows and associated energy requirements.

Material demand as a function of economic activity or bottom-up human activities. A key principle for energy-economy models is to explicitly describe material demand (in physical quantities) as function of economic activity. In this work, we have done so by a simplistic (although consistent with historical trends) approach, linking global GDP to mineral demand. Other approaches may include the use of econometric techniques linking particular socio-economic drivers (e.g. population, sectoral output) to material demand, or the use of material intensities for each economic sector. Material demand may also be estimated by directly quantifying the material requirements of human activities, i.e. translating an explicit representation of services such as transportation, housing, infrastructure into material requirements, which is the approach increasingly taken by the IMAGE model. Particular attention should be given to the continuity with historical trends, and in the case of an important decoupling of economic activity and material demand occurring, the underlying socio-economic drivers should be made explicit in the modelling.

*Explicitly modelling in-use stocks and flows.* We have shown that the extent of recycling is critical when determining the energy consumption of mining activities. However, the extent of recycling is determined not only by end-of-life recycling rates, but also by the mineral material flows available for recycling at a given time, which are function of the share of in-use stocks reaching their end-of-life, and hence, of the lifetimes of each material in society (Busch et al., 2014). Recent work by Elshkaki et al. (2018) and Deetman et al. (2021) explicitly models in-use stocks and shows that the availability of materials acts as an important limiting

<sup>&</sup>lt;sup>13</sup> Examples include galvanisation and sacrificial anodes (e.g. zinc, magnesium, aluminium), pigments (titanium, cobalt, bismuth), fertilizers and pesticides (e.g. phosphorous, copper, selenium), additives in (petro) chemicals (e.g. platinum to improve the combustion of gasoline), use in catalysis (e.g. platinum group metals, germanium), pyrotechnics and fireworks (e.g. aluminium, copper, chromium). For a broader review of dispersive uses of metals, see Ciacci et al. (2015).

factor to the potential of recycling. Such explicit representation of in-use stocks and end-of-life flows is crucial as it prevents from modelling an extent of recycled materials inconsistent with physical stocks.

Explicitly modelling the energy requirements associated with mineral material flows. It seems crucial that the energy requirements associated with mineral material flows (mining and refining, and secondary recycling) are explicitly represented in models. Such energy requirements may be quantified through an increase in the material producing sectors output (and consequently an increase in their energy consumption), or through the use of energy intensities of production for each material. The former technique may require translating a demand for heterogeneous materials into a demand for sectors aggregating numerous materials, and to use monetary units, which may distort physical flows and their energy requirements. Conversely, the latter approach is likely to remain closer to underlying physical flows, but care should be given to the evolution of energy intensities of production over time, so that they account both for thermodynamic limits (i.e. limit to efficiency gains that can be achieved), and for the increasing energy intensities of mining activities - an explicit modelling of the mining extraction stage may hence be helpful.

*Mineral materials to consider.* The number of mineral materials that can be realistically modelled is necessarily limited by time and resources available. This study has allowed us to explore the magnitude of the mining industry's historical final energy consumption related to different minerals, as well as potential evolutions in the future. Hence, we suggest that in addition to steel and cement, which are traditionally considered in energy-economy models, other relevant mineral materials responsible for a high final energy consumption include aluminium, copper, gold, limestone, sand and gravel (partly considered as cement), and silver. In addition, other mineral materials may be worth modelling for instance due to significant energy requirements in the mineral refining stages, or for different reasons such as criticality for the energy transition and supply risks (Valero et al., 2018).

#### 7. Conclusion

This paper has provided an estimate of the historical final energy consumption of the mining industry globally, as well as an exploratory analysis of future possible pathways for the mining industry's final energy consumption (excluding fossil fuel extraction activities). We find that the mining industry is currently responsible for a small, and yet significant, share of global final energy consumption - approximatively 1.7%. However, such a share is likely to increase considerably in the future as a result of a substantial increase in the mining industry's final energy consumption if current trends continue (i.e. high economic growth alongside a high material-GDP coupling), until reaching a value in the range 4–12% of forecasted global final energy consumption for the socioeconomic scenarios adopted in this study. We also find that the mining industry's future final energy consumption is first and foremost determined by future economic activity: final energy consumption due to mineral demand for energy transition technologies is dwarfed by the final energy consumption due to mineral demand for the rest of the economy. In addition, future recycling rates and energy intensities of mining are key factors determining the mining industry's future final energy consumption - while the latter is partly exogenous (due to geological constraints), the former is dependent on political, industrial, and technological choices.

This study has found that mineral material flows and associated energy consumption are only covered to a limited extent in the energyeconomy models reviewed. We argue that mineral material flows need to be explicitly represented for a set of critical materials, from the mineral mining stage to the mineral refining and recycling stages, and that the energy implications of such flows need to be explicitly modelled, so that models produce internally consistent scenarios. Particularly, it is crucial that models explicitly represent (i) material demand as function of economic activity or underlying human activities and services, (ii) primary mineral extraction and material recycling as function of material demand, end-of-life materials, and end-of-life recycling rates, and (iii) the energy requirements of these material flows and processing activities, taking into consideration the increasing energy intensities of mining activities due to mineral depletion.

Lastly, our results, combined with the limited coverage of material flows in energy-economy models, raises concerns regarding the consistency of mainstream socio-economic scenarios in terms of relationship between economic activity and final energy consumption forecasts. Indeed, when tightly coupling material demand to economic activity, consistently with historical trends, we find that the mining industry's final energy consumption may increase considerably and account for a significant fraction of global final energy consumption, hence raising the concern that global final energy consumption may be underestimated in mainstream socio-economic scenarios. The limited consideration of mineral material flows and associated energy requirements seems to be an important blind spot in energy-economy models and may hinder the efforts of the community to build consistent energy transition pathways.

#### CRediT authorship contribution statement

**Emmanuel Aramendia:** Conceptualization, Methodology, Software, Validation, Formal analysis, Data curation, Writing - original draft, Writing - review & editing, Visualization. **Paul E. Brockway:** Conceptualization, Methodology, Validation, Writing - original draft, Writing review & editing, Supervision, Funding acquisition. **Peter G. Taylor:** Conceptualization, Methodology, Validation, Writing - original draft, Writing - review & editing, Supervision. **Jonathan Norman:** Conceptualization, Methodology, Validation, Supervision.

#### **Declaration of Competing Interest**

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

#### Data availability

The R code that was used for this analysis can be found in the online repository associated with this article at the University of Leeds Data Repository: https://doi.org/10.5518/1420. Input data, including recycling rates and energy intensities of mining, can be found in the Supplemental Information 2. Future recycling rates and  $\alpha$  values can also be found in Supplemental Information 2. Input data in the format used by the R code can be found in the online repository, except mineral extraction data and International Energy Agency data, which are proprietary. Further data may be available upon reasonable request to the authors.

#### Acknowledgments

We would like to thank Lina Brand-Correa for feedback on early versions of this work, as well as Olivier Vidal, Baptiste Andrieu, and François Verzier for discussions and feedback on the article. We also would like to thank Fridolin Krausmann for providing historical bulk material production data, Iñigo Capellán-Pérez, for providing future material consumption data due to the energy transition according to different scenarios run by the MEDEAS Integrated Assessment Model as well as helpful feedback, and Massimo Pizzol, for helpful discussion on the Life Cycle Analysis methodology. Last, we would like to thank the modelling teams that responded our questions regarding the different models (IEA's World Energy Model, AIM/CGE, GCAM, IMAGE, Shell's World Energy Model, REMIND-MagPIE, E3ME, U.S. EIA's WEPS, GEM-

E3/PRIMES, MEDEAS, POLES, and MESSAGE). The contributions of P. G.T. and J.N. were supported by the Centre for Research into Energy Demand Solutions funded by UK Research and Innovation (grant number EP/R035288/1). We also acknowledge support for P.E.B. under EPSRC Fellowship award EP/R024251/1, and for E.A., funded by the School of Earth of Environment of the University of Leeds, in support of P.E.B.'s fellowship award.

#### Appendix A. Glossary of energy terms

Table A.1 introduces a short glossary of the energy-related terminology used throughout the paper.

#### Table A.1

Glossary of energy-related terminology used throughout the article.

Term	Definition
Direct energy	Final energy used in situ by an industry, or in general, by any system (e.g. plant, city, country, etc.). May be quantified in terms of primary or final
	energy.
Energy intensity	When mentioned in relationship to the work conducted in this study, equivalent to final energy intensity.
Final energy	Energy used by the end-user in the form of a final energy carrier (e.g. electricity, gasoline, etc.)
Final energy intensity	Final energy required to mine a unit of a given mineral, including both the direct and indirect energy requirements. It is calculated multiplying the
	GER by the final-to-primary energy ratio calculated for the mining industry. (See SI-1.)
Future energy intensity	Final energy required to mine a unit of a given mineral for the future analysis, evolves dynamically following one of the three scenarios defined in
	Section 3.2.4.
Gross Energy Requirements	Total of direct and indirect energy requirements associated with a system (in the paper's case, the mining of one unit of a given mineral), quantified in
(GER)	terms of primary energy requirements.
Historical energy intensity	Final energy required to mine unit of a given mineral for the historical analysis, determined using the literature (see SI-2), and kept constant over the
	timespan of the historical analysis (1971–2015).
Indirect energy	Final energy used ex-situ, i.e. final energy used in the supply chain, of an industry, or in general, of any system (e.g. plant, city, country, etc.). May be
	quantified in terms of primary or final energy.
Primary energy	Energy flows extracted from the environment (e.g. crude oil, wind power, solar radiation, etc.)
Primary energy intensity	Primary energy required to mine a unit of a given mineral, equivalent to the Gross Energy Requirements of mining a unit of a given mineral.

#### Appendix B. The mineral extraction process

The subsequent summary of the mineral extraction process is based on [Rankin, 2011, Chap. 7 and 8], which provides an excellent overview of the different extraction processes.<sup>14</sup>

*Ore extraction.* The ore of interest is extracted from the ground, lifted and hauled until the processing facility, where it is further processed.<sup>15</sup> The extracted ores that are not of sufficient economic interest constitute the overburden, and are left unexploited in the mining site.

Ore beneficiation. The extracted ore is then concentrated. The beneficiation (or concentration) process consists of separating the mineral of interest from the other minerals, i.e. the *gangue*. Usually, the process involves crushing and grinding the ore (process known as ore *comminution*), until reaching the ore's *liberation size*, at which the particles of interest are released from the rest of the ore. The ore concentrate is then obtained by separating the mineral of interest from the gangue, which ends up as *tailings*, i.e. processed "waste" ore. The ore concentrate is an ore consisting mostly of the mineral of interest, but in which considerable impurities remain.

*Concentrate refining.* It is usually the case for metals that the ore concentrate previously obtained needs to be purified.<sup>16</sup> Different techniques can be used to refine the concentrate and obtain the final product. These techniques can be classified in terms of pyrometallurgical and hydrometallurgical treatments. When using pyrometallurgical treatments, the ore concentrate is smelted at high temperatures (typically in the range 500–2000°C), often in the presence of a reducing agent (for instance coke for iron ore concentrate), to obtain the final metal. When using hydrometallurgical treatments, the metal is extracted either from the extracted ore or from the beneficiated ore using a liquid substance (i.e. the *lixiviant*) – operation known as leaching. Then, the solution may be refined with a set of techniques, including chemical precipitation, solvent extraction, or electrowinning. Finally, the metal needs to be recovered, which can also be done through different techniques, such as electrowinning.

#### Appendix C. Increasing energy intensities: a comparison

Fig. C.1 shows the historical evolution of copper mining energy intensities in Chile (2001–2019), which has increased by 66% over the covered time period, or by a yearly rate of 3%. The rest of this appendix compares the energy intensity scenarios developed in this study with those obtained in the literature.

<sup>&</sup>lt;sup>14</sup> Regarding energy consumption, we note that the prevalent extraction stage depends on the mineral extracted as well as on the extraction route. See e.g. Norgate et al. (2011) and Nuss et al. (2014) for a breakdown for different minerals.

<sup>&</sup>lt;sup>15</sup> This step is skipped when the technique of *in situ* mineral leaching is used to directly recover the mineral of interest.

<sup>&</sup>lt;sup>16</sup> Some minerals that do not need to be purified, for instance some construction and industrial minerals (sand, gravel...), are only likely to undergo the two first steps.



Fig. C.1. Energy intensity of copper mining over time in Chile. Time series deduced from the data provided by the Chilean Copper Commission. (See e.g. (Chilean Copper Commission, 2020)).

Table C.1 summarises the projections of energy intensities of mining that are used by other studies. The increase factor refers to  $a_{m,t}$  in Eq. 6. To compare, the increase factors we use reach 1.42 in 2050 in the low increase scenario, and reach 2.36 in 2050 in the high increase scenario (it remains constant and equal to unity in the no increase scenario). Hence, the factors we use, particularly in the case of the low increase scenario, are reasonable when compared to factors in Table C.1. The high increase scenario has higher increasing energy intensity factors than most of these studies, but is rather in the lower end of factors reported in Harmsen et al. (2013). We note that the range of increasing energy intensities reported by Harmsen et al. (2013) places the study as an outlier, and based on our judgment and discussion with external experts (see acknowledgments), the study was not used to inform the energy intensities scenarios considered in each study, which lead to different cumulative primary extraction (because of e.g. differences in demand as well as recycling rates), and hence to different energy intensities in 2050; (ii) to the consideration or not of technological improvements, (iii) to uncertainties in the ore grade-tonnage curves. We note that two studies (Kuipers et al., 2018; Dong et al., 2020) are based on van der Voet et al. (2019) and have therefore similar results.

#### Table C.1

Summary of increasing energy intensities factors in other studies, deduced from the information available (equations, graphs, etc), so values may not be accurate. Increase factors are noted NA for Ciacci et al. (2020) because the values are scenario dependent, and are not clearly stated in the article, although the methodology is the same than for Elshkaki et al. (2016), but with different scenarios.

Study	Mineral	Region	Time period	Increase factor	Methodology
Ester Van der Voet et al. (2019)	Copper Zinc Lead Nickel sulfides Nickel laterites	World	2010–2050	1.30 1.35 1.75 1.12 1.14	Ore grades are extrapolated as a function of time fitting historical data. Then an empirical relationship linking ore grades to energy intensities is applied.
Kuipers et al. (2018)	Copper	World	2010-2050	1.30	Same method and values as (Ester Van der Voet et al., 2019)
Dong et al. (2020)	Copper	China	2010-2050	1.42	Same method as (Ester Van der Voet et al., 2019), different values (specific to China).
Elshkaki et al. (2016)	Copper pyro Copper hydro	World	2010–2050	1.33–2.00 1.36–2.01	Ore grade is determined for each scenario as a function of cumulative extraction, and a energy consumption is determined as function of ore grade.
Ciacci et al. (2020)	Copper	EU-28	NA	NA	Same approach and values as (Elshkaki et al., 2016).
Harmsen et al. (2013)	Copper	World	2010-2050	2.54-6.82	Ore grade is determined for each scenario as a function of cumulative extraction, and a energy consumption is determined as function of ore grade.

#### Appendix D. Ore grade-tonnage distributions

Fig. D.1 shows the unimodal and bimodal ore grade-tonnage distribution curves (tonnage as function of ore grade), as well as their implications in terms of energy intensities of mining as function of ore grades – the grey area represents the tonnage extracted to date.



Fig. D.1. Unimodal and bimodal ore grade-tonnage distribution curves, as well as implications of each distribution in terms of evolution of energy intensities as function of ore grades. The grey area corresponds to the tonnage extracted to date.

According to Skinner (1976), in the case of the unimodal distribution, common rocks are constituted by different minerals, "one or more of which contains a geochemically abundant metal as an essential constituent" (Skinner, 1976). The consequence is that producing a mineral concentrate in which a given geochemically abundant element is the main component is possible, without needing to break all minerals to liberate the atoms of interest. Conversely, in the case of the bimodal distribution, only the small distribution peak corresponds to mineralogical deposits where the element of interest is present in mineral compounds of its own, which can be concentrated. Instead, the high peak at low concentrations corresponds to the element being present in minerals "as randomly distributed atoms trapped by isomorphous substitution in minerals of the geochemically abundant element replacing an atom of geochemically abundant element" (Skinner, 1976). The consequence is that minerals need to be broken down to liberate and concentrate the elements of interest, which translates into very a high energy consumption – much higher than in the case of elements being available in minerals of their own.

The consequence is that energy intensities increase in a continuous way when ore grade decreases in the case of the unimodal distribution, so that the historical relationship between ore grade and energy intensity is likely to continue over time. Conversely, in the case of the bimodal distribution, the energy intensity increases steeply at a given ore grade resulting from the transition from mineral deposits where the element of interest is present in minerals of its own, to deposits where the element of interest is only available as a randomly distributed element substituting more abundant elements. Hence, when reaching ore grades somewhere between the small and high peaks of the ore grade-tonnage distributions, a mineralogical barrier is reached, at which grade energy intensities steeply increase, which may prevent, or strongly limit, extraction of ores with ore grades lower than the critical mineralogical barrier ore grade. Skinner (1976) defends that of industrial metals, only aluminium, iron, magnesium, manganese, and

titanium are likely to follow the unimodal distribution curve. However, there are large uncertainties on the exact shape of distribution curves for each element, and Arndt et al. (2017) explains for instance that the unimodal distribution currently appears to be more likely in the case of copper although it is a geochemically scarce metal – because it is found as minerals of its own even at low concentrations.

To conclude, there are large uncertainties regarding the distribution curves that may be followed by each element, which translates in large uncertainties on the future evolution of energy intensities, which are intertwined with ore grade-tonnage distributions.

#### Appendix E. Data used

#### E.1. Scenario data

We take final energy consumption and GDP data for the SSP1, SSP2, SSP5, LED and B2DS socio-economic scenarios, used in the IPCC's Special Report on 1.5°C [(Masson-Delmotte et al., 2018, Chap. 2)], from the 1.5°C Scenario Explorer (Huppmann et al., 2019). For the post-growth socioeconomic scenario, we take global GDP declining by 0.2% each year, and we determine the yearly final energy consumption using a linear regression of global final energy consumption as function of global GDP.

#### E.2. Historical production data

Historical data of mineral production (1970–2015) are taken for most minerals from the United States Geological Survey (Kelly and Matos, 2016). For uranium production, data are taken from Grancea and Hanly (2018) (p. 89), and for clays, sand and gravel, and limestone, from Krausmann et al. (2018), who kindly provided the data on our request. It is worth noting that the data we use from the United States Geological Survey is constructed by combining country and company level data, so that like data from the IEA, it is likely to be missing informal mining activities - although it is likely to be more comprehensive than data from the IEA as it also uses company level data (the IEA only uses country level reporting). Next, the data we use from the work of Krausmann et al. (2018) is a global estimation of mineral extraction using a wide range of methods depending on the mineral, which does not rely on official primary extraction data, and hence does not suffer from the same drawbacks.

#### E.3. Initial recycling rates

Recycling rates in terms of recycled content are taken from Graedel et al. (2011) (Appendix C) for metals (when more than one value is reported, we use the average of the lowest and highest value), and from Haas et al. (2005) (Table S1) for other minerals, using the average values reported for industrial minerals and construction minerals.

#### E.4. Historical demand data

The historical demand for each mineral m is determined from its historical production  $P_m$  and recycling rate  $r_m$  following:

$$D_m = \frac{P_m}{1 - r_m}.$$
(E.1)

#### E.5. Historical energy intensities

The initial GERs (which represent the primary energy intensities) of mineral extraction are derived from the work of Norgate et al. (2011); Nuss et al. (2014); Norgate et al. (2012); Rankin (2011); Mudd (2012); Hammond et al. (2011); and Calvo et al. (2018), and reported in SI-2, alongside comments regarding assumptions for their estimation - assumptions were sometimes needed when the literature did not allow to obtain directly the GER associated to mining, for instance when the GER reported included the energy requirements of both the mining and the metallurgical steps without breakdown. Then we multiply the GER by the average final-to-primary energy ratio for the mining and quarrying industry globally, that is determined using data from the International Energy Agency and a recently developed Physical Supply Use Table framework for the energy industry (Heun et al., 2018; Aramendia et al., 2022), which yields the final energy intensity. SI-1 explains briefly the methodology for the calculations using the Physical Supply Use Table framework.

#### E.6. Projections of mineral demand for Energy Transition Technologies

The mineral requirements reported in Watari (2020) are available as supplemental information of the cited paper. The mineral requirements of the energy transition scenarios conducted in Capellán-Pérez et al. (2019) are shared in the online repository associated to this paper, with agreement of the authors of the study.

#### Appendix F. Linear regression examples

Fig. F.1 shows the linear regressions conducted for 8 of the main minerals to assess the mineral requirements of the rest of the economy following Eq. 3. If we look at the subset of minerals for which future demand for the rest of the economy is determined with a linear regression (conversely to those that do not correlate well with GDP and that are hold constant), the 8 minerals included in Fig. F.1 always add up to at least 85% of the final energy consumption of the subset. Hence, the quality of the fits obtained validate the method introduced in Section 3.2.1.

)



Fig. F.1. Examples for 8 main minerals of the linear regressions conducted to determine the mineral requirements of the rest of the economy (Eq. 3). These 8 minerals always account for 85% of the final energy consumption of extraction, for the subset of minerals which mineral requirements by the rest of the economy are determined with a linear regression.

We note that for some minerals (aluminium, iron ore, nickel, zinc), the recent trend is steeper than the long-term trend. Thus, we conduct a sensitivity analysis in which we fit the linear regression for the 1995–2015 period for all minerals. Results for this sensitivity analysis are available in SI-1; conclusions and trends obtained in the main paper remain unchanged, although the projected mining industry's final energy consumption reaches values somewhat higher (approximatively, by 10–15% higher, depending on the scenario) when conducting the fit in the most recent time period. However, and as discussed in Section 3.3.2, such extrapolation of historical trends does not take into consideration structural changes that may allow material demand to be partly decoupled from economic activity.

Then, Table F.1 gives the values of coefficients  $a_m$ ,  $b_m$ , and  $r^2$ , and specifies the fitting time period (1971–2015 or 1990–2015) for each mineral.

Table F.1

Values of  $a_m$ ,  $b_m$ , and  $r^2$ , and fitting time period. Mt: 10<sup>6</sup> tons. TUS\$: 10<sup>9</sup> US\$. US\$ in constant 2010 values. Dashes refer to non applicable values (minerals for which demand is hold constant and equal to  $b_m$ ). Dmnl: dimensionless.

Mineral <i>m</i>	Fitting period	$a_m$	$b_m$	р	$r^2$
		Mt/TUS\$	Mt	Dmnl	%
Aluminium	1971–2015	1.10	-1.06e1	3.70e-27	0.929
Antimony	1971-2015	3.02e-3	-1.94e-3	2.57e-16	0.786
Arsenic	_	_	3.37e-2	_	_*
Asbestos	_	_	2.15	_	-
Barite	1995–2015	8.50e-2	2.25	8.97e-06	0.655
Beryllium	_	_	2.69e-4	_	-
Bismuth	1995–2015	2.26e-4	-6.75e-3	7.21e-08	0.790
Boron	1971-2015	1.09e-1	1.15	9.46e-16	0.773
Cadmium	1971-2015	1.86e-4	3.06e-2	3.32e-10	0.602
Chromium	1995–2015	2.18e-1	-5.11	4.88e-11	0.902
Clays	_	_	5.62e + 3	_	-
Cobalt	1995–2015	3.18e-3	-1.05e-1	2.90e-14	0.955
Copper	1971-2015	3.16e-1	1.73	6.22e-40	0.982
Diatomite	1971-2015	1.73e-2	1.17	3.58e-13	0.703
Feldspar	1971-2015	4.39e-1	-8.91	2.55e-29	0.945
Fluorspar	1995–2015	1.12	-7.43e-1	4.75e-07	0.745
Gallium	1995–2015	1.65e-5	-7.17e-4	1.02e-05	0.650
Germanium	1995–2015	6.39e-6	-2.06e-4	1.62e-10	0.889
Gold	1971-2015	4.62e-5	9.21e-4	6.79e-19	0.836
Graphite	1971-2015	1.43e-2	1.59e-1	5.66e-18	0.820
Gypsum	1971-2015	3.52	-2.73e + 1	2.14e-20	0.860
Indium	1971-2015	1.28e-5	-3.01e-4	1.40e-22	0.888
Iron ore	1971-2015	4.08e + 1	1.67e + 2	2.61e-17	0.807
Kyanite	-	-	3.98e-1	_	-
Lead	1995–2015	1.57e-1	-1.31	3.19e-09	0.848
Limestone	1971-2015	8.58e + 1	-1.33e + 3	1.38e-26	.927
Lithium	1995–2015	1.47e-2	-4.99e-1	3.59e-11	0.905
Magnesium	1995–2015	2.89e-2	-7.73e-1	5.88e-13	0.938
Manganese	1995–2015	5.36e-1	-1.35e + 1	2.11e-11	0.910
Mercury	-	-	3.26e-3	-	-

(continued on next page)

#### E. Aramendia et al.

#### Table F.1 (continued)

Mineral <i>m</i>	Fitting period	$a_m$	$b_m$	р	$r^2$
		Mt/TUS\$	Mt	Dmnl	%
Molybdenum	1995-2015	7.15e-3	-1.40e-1	5.96e-10	0.873
Nickel	1971-2015	4.69e-2	-1.97e-1	1.19e-20	0.864
Niobium	1995-2015	2.21e-3	-7.11e-2	1.03e-05	0.650
Perlite	1970-2015	4.69e-2	2.92e-2	1.26e-11	0.651
Phosphate rock	1995-2015	3.22	-2.29e + 1	3.32e-08	0.807
Platinum group metals	1970-2015	1.11e-5	4.84e-5	5.25e-23	0.893
Potash	1995-2015	4.22e-1	5.97	6.78e-06	0.664
Pumice and pumicite	-	_	1.79e + 1	-	-
Rare Earth Elements	1970-2015	2.17e-3	-2.68e-2	3.80e-24	0.905
Rhenium	1995-2015	8.08e-7	2.24e-6	3.97e-07	0.750
Salt	1970-2015	2.41	9.88e + 1	3.10e-31	0.955
Sand and Gravel	1970-2015	4.93e + 2	-1.94e + 3	5.75e-31	0.954
Selenium	1970-2015	1.99e-5	9.02e-4	2.99e-12	0.673
Silicon	1995-2015	1.78e-1	-4.81	2.93e-15	0.965
Silver	1970-2015	3.75e-4	5.21e-3	9.78e-34	0.965
Soda ash	1995–2015	7.37e-1	1.40e-1	2.92e-17	0.978
Sodium sulfate	1970-2015	4.05e-2	3.56	2.21e-17	0.808
Strontium	_	_	3.93e-1	-	-
Talc and Pyrophyllite	_	_	8.11	-	-
Tantalum	1970-2015	2.34e-5	-1.94e-4	5.23e-12	0.665
Tin	_	-	3.28e-1	-	-
Titanium mineral concentrates	1970-2015	1.26e-1	1.20	3.14e-30	0.950
Tungsten	1995–2015	2.96e-3	-6.59e-2	1.63e-11	0.913
Uranium	_	_	5.11e-2	_	-
Vanadium	1995-2015	1.46e-3	-2.93e-2	3.12e-14	0.955
Vermiculite	_	-	4.37e-1	-	-
Wollastonite	_	-	6.07e-1	-	-
Zinc	1970-2015	1.86e-1	2.53	8.54e-31	0.953
Zirconium	1970-2015	1.97e-2	7.16e-2	1.91e-19	0.845

#### Appendix G. Energy-economy models review: details

Table G.1 introduces each reviewed model alongside the main references, and summarises the findings of our review following for the four following criteria: (i) mineral materials covered, (ii) description of material demand, (iii) description of primary mineral extraction, and (iv) feedback of material flows on energy consumption – note that no model explicitly represented increasing energy intensities of mining activities.

#### Table G.1

Performance of each reviewed energy-economy model for four criteria: (i) mineral materials covered, (ii) description of material demand, (iii) description of primary mineral extraction, and (iv) feedback of material flows on energy consumption. No model explicitly represents the increasing energy intensities of mining activities. By *energy intensities of production*, we mean a value linking physical production to energy consumption, expressed in a unit such as GJ/tonne. WEPS: World Energy Projection System. ETTs: Energy Transition Technologies. WEM: World Energy Model. REEs: Rare Earth Elements. PGMs: Platinum Group Metals. Note: other materials, such as fossil fuels, plastics, wood, food, are excluded from this review.

Model	Ref	Materials covered in physical units	Determination of material demand	Primary and secondary production	Feedback on energy consumption
IEA's WEM	IEA (2021)	Steel, aluminium, copper, nickel, lithium, cobalt, REEs. Only clean technology sector: zinc, PGMs, manganese, graphite, molybdenum.	Demand split by (i) uptake of clean technologies for the energy transition, and (ii) relevant activity drivers (GDP, industry added value, population) for the rest of the economy.	Primary and secondary production differentiated using in-use stocks and end-of-life recycling rates.	Only for steel and aluminium, through the use energy intensities of production, which decrease over time to model increases in efficiency.
AIM/CGE	Fujimori et al. (2017)	Steel, cement	Linear relationship with sectoral output.	No explicit differentiation.	Modelled through the increase in sectoral output.
IMAGE	Stehfest et al. (2014)	Steel, cement	Increasingly through a detailed representation of human activities and services (housing, mobility, infrastructure, etc. requirements translated into material requirements). Use of GDP per capita and population in some cases.	Differentiation for steel, using in- use stocks and end-of-life recycling rates.	Yes, through the use of energy intensities of production, which decrease over time to model increases in efficiency.
Shell's WEM	Shell (2017)	Aggregated in four groups: (1) iron and steel, (2) non-ferrous metals, (3) non-metallic minerals, (4) glass.	Using economic data (GDP per capita) and different evolutions depending on scenario narratives.	No explicit differentiation.	Yes, through the use of energy intensities of production, which decrease over time to represent both increases in efficiency and in the recycling of metals.

(continued on next page)

## Table G.1 (continued)

Model	Ref	Materials covered in physical units	Determination of material demand	Primary and secondary production	Feedback on energy consumption
REMIND- MAgPIE	Baumstark et al. (2021)	Steel, cement	Driven by population and GDP, decrease of demand with carbon pricing.	Steel is differentiated using an external steel stock model, with end-of-life recycling rates. Because the model is external, there is no interaction with the scenario and the availability of secondary steel. All cement comes from primary production.	General equilibrium model: steel and cement demand influences final energy demand, and prices of final energy carriers influence back demand for steel and cement, until equilibrium is found.
E3ME	Cambridge Econometrics (2019)	Aggregated in four groups: (1) construction minerals, (2) industrial minerals, (3) ferrous metals, (4) non-ferrous metals.	Determined through econometric equations, as function of gross economic output by sector, material prices, and innovation.	Materials-related policies can be implemented so that the demand for virgin materials are decreased, and the demand for recycled materials are increased, through exogenous recycled contents.	Demand for materials feedbacks on the gross output of material- producing sectors, which then feedbacks on the final energy demand of those sectors. When more recycled materials are demanded, the gross output of the recycling sector increases while the gross output of virgin material producer sectors decreases.
EIA's WEPS	EIA (2021)	Steel	Monetary demand for the iron and steel sector is determined by the economic module, and then converted in physical units (tonnes) following historical trends.	Differentiation of primary and secondary production following historical trends, work ongoing to represent constraints on scrap availability.	Yes, energy consumption by fuel determined as function of primary and secondary steel production and of the manufacturing technologies used.
GCAM	Bond-Lamberty et al. (2021)	Cement	Function of GDP.	No explicit differentiation.	Yes, through the use of energy intensities of production, which decrease over time to model increases in efficiency.
IMACLIM – national versions	(Le Treut, 2018; Le Treut, 2020)	Cement and steel on the French version of the model, none in other versions.	Demand for cement and steel is determined within the general equilibrium using prices and elasticities (price and income), with consideration of a minimum level of final demand needed to provide basic needs (set as exogenous parameters).	No differentiation between primary and secondary production	Energy feedback through the general equilibrium, either directly or alternatively, using the total sectoral monetary output determined by the general equilibrium, and exogenous sectoral energy intensities provided by the modeller.
GEM-E3/ PRIMES	E3Modelling (2017); E3Modelling, 2018	Steel, metal products, non- metallic minerals (including bricks, ceramics, glass, sand, cement)	Monetary demand for each sector (provided by GEM-E3) is translated in PRIMES to a physical demand for each mineral material (in tonnes) using material intensities (tonnes per monetary output) for each sector.	Differentiation of primary and secondary production for steel and other relevant minerals. No consideration of in-use stocks and scrap availability.	Yes, a technology mix is determined through a cost optimisation procedure depending on climate policies and prices, which gives an energy intensity of production for each mineral material.
MEDEAS	Capellán-Pérez et al. (2019); Capellán-Pérez et al. (2020)	Numerous materials at least partially covered, some only for energy transition technologies (most abundant ones), and some also for the rest of the economy (scarcest ones).	Demand split by (i) uptake of energy transition technologies, and (ii) the rest of economic activities, with mineral demand as a linear function of GDP.	Materials-related policies can be implemented so that the demand for virgin materials are decreased, and the demand for recycled materials are increased, through exogenous recycled contents. Consideration of in-use stocks and scrap availability only for ETTs.	Only partially and indirectly, through the energy requirements of energy transition technologies, which modify the Energy Return On Investment of the energy system, and then feedback on global final energy demand.
POLES	Després et al. (2018)	Only steel, although previous studies looked at cement, copper, aluminium, and glass.	Current version: demand for steel is mostly determined from economic activity data, i.e. using a material intensity in tons/GDP for each country. Version under development: demand for specific end-uses (automotive, building, power sectors) is calculated using bottom-up activity data (i.e. number of	Differentiation of primary and secondary steel using the availability of steel scrap at a given time, which is function of the in-use stocks of steel in each type of equipment, of the lifetime of each equipment, and of the end-of-life recycling rates of scrap steel. Primary production covers remaining demand.	Yes, through the use of energy intensities of production, which vary as function of energy prices. Version under development: production processes broken down in different processes, each with own intensity of production, with the share of each production process determined through a cost optimisation procedure.

(continued on next page)

#### Table G.1 (continued)

Model	Ref	Materials covered in physical units	Determination of material demand	Primary and secondary production	Feedback on energy consumption
			vehicles produced multiplied by steel required for a vehicle).		
MESSAGE	Krey et al. (2020)	Steel, Cement, Aluminium	Demand determined bottom-up for major end-uses (buildings and transports). Rest of demand is determined using an econometric formulation, as function of GDP and population.	A share of production becomes scrap at each time. Consideration of in-use stocks, end-of-life materials and scrap availability only for electricity generation technologies. Secondary production driven by scrap availability and costs compared to primary production.	Material demand is satisfied by particular production technologies (mix determined following a cost minimisation procedure), which each have a specific energy intensity.

#### Appendix H. Supplementary data

Supplementary data associated with this article can be found, in the online version, at https://doi.org/10.1016/j.gloenvcha.2023.102745.

#### References

- Abraham, David S., 2015. The elements of power: Gadgets, guns, and the struggle for a sustainable future in the rare metal age. Yale University Press.
- Aitken, Douglas, Rivera, Diego, Godoy-Faundez, Alex, Holzapfel, Eduardo, 2016. Water Scarcity and the Impact of the Mining and Agricultural Sectors in Chile. Sustainability 8 (2), 128. https://doi.org/10.3390/su8020128.
- Allwood, Julian M., Ashby, Michael F., Gutowski, Timothy G., Worrell, Ernst, 2013. Material efficiency: providing material services with less material production. Philos. Trans. R. Soc. A Math. Phys. Eng. Sci. 371 (1986), 1471–2962. https://doi.org/ 10.1098/rsta.2012.0496.
- Anderson, K., Peters, G., 2016. The trouble with negative emissions. Science 354 (6309), 182–183. https://doi.org/10.1126/science.aah4567.
- Aramendia, Emmanuel, Heun, Matthew K., Brockway, Paul E., Taylor, Peter G., 2022. Developing a Multi-Regional Physical Supply Use Table framework to improve the accuracy and reliability of energy analysis. Appl. Energy 310. https://doi.org/ 10.1016/j.apenergy.2021.118413.
- Arndt, Nicholas T., Fontbote, Lluis, Hedenquist, Jeffrey W., Kesler, Stephen E., Thompson, John F.H., Wood, Daniel G., 2017. Future global mineral resources. Geochemical. Perspectives 6 (1), 1–171. https://doi.org/10.7185/ geochempersp.6.1.
- Bardi, Ugo, 2013. The Mineral Question: How Energy and Technology Will Determine the Future of Mining. Front. Energy Res. 1 https://doi.org/10.3389/ fenrg.2013.00009.
- Bardi, Ugo, 2014. Extracted: How the quest for mineral wealth is plundering the planet. Chelsea Green Publishing. ISBN 1-60358-541-9.
- Baumstark, Lavinia, Bauer, Nico, Benke, Falk, Bertram, Christoph, Bi, Stephen, Gong, Chen Chris, Dietrich, Jan Philipp, Dirnaichner, Alois, Giannousakis, Anastasis, Hilaire, Jerome, Klein, David, Koch, Johannes, Leimbach, Marian, Levesque, Antoine, Madeddu, Silvia, Malik, Aman, Merfort, Anne, Merfort, Leon, Odenweller, Adrian, Pehl, Michaja, Pietzcker, Robert C., Piontek, Franziska, Rauner, Sebastian, Rodrigues, Renato, Rottoli, Marianna, Schreyer, Felix, Schultes, Anselm, Soergel, Bjoern, Soergel, Dominika, Strefler, Jessica, Ueckerdt, Falko, Kriegler, Elmar, Luderer, Gunnar, 2021. REMIND2.1: Transformation and innovation dynamics of the energy-economic system within climate and sustainability limits. Geosci. Model Dev. 14 (10), 6571–6603. https:// doi.org/10.5194/gmd-14-6571-2021.
- Beylot, Antoine, Guyonnet, Dominique, Muller, Stephanie, Vaxelaire, Stephane, Villeneuve, Jacques, 2019. Mineral raw material requirements and associated climate-change impacts of the French energy transition by 2050. J. Clean. Prod. 208, 1198–1205. https://doi.org/10.1016/j.jclepro.2018.10.154.
- Bihouix, Philippe, De Guillebon, Benoit, 2012. Quel futur pour les metaux?: Raréfaction des métaux: un nouveau défi pour la société. EDP Sci. ISBN 2-7598-0901-3.
- Bithas, Kostas, Kalimeris, Panos, 2018. Unmasking decoupling: Redefining the Resource Intensity of the Economy. Sci. Total Environ. 619–620, 338–351. https://doi.org/ 10.1016/j.scitotenv.2017.11.061.
- Bleischwitz, Raimund, Nechifor, Victor, Winning, Matthew, Huang, Beijia, Geng, Yong, 2018. Extrapolation or saturation – Revisiting growth patterns, development stages and decoupling. Glob. Environ. Chang. 48, 86–96. https://doi.org/10.1016/j. gloenvcha.2017.11.008.
- Bond-Lamberty, B., Patel, P., Lurz, J., Smith, S., abigailsnyder, pkyle, kvcalvin, Kalyn R. Dorheim, Robert Link, mbins, skim301, Aaron S., Leyang Feng, Sean W D Turner, cwroney, Cary Lynch, Jhoring, Zarrar Khan, Haewon, mwisepnnl, mollycharles, Gokul Iyer, Alexey Shiklomanov, swaldhoff, Richard Plevin, matteomuratori, amundra, Corinne Hartin, and Kanishka Narayan. Jgcri/gcam-core: GCAM 5.4. 2021. https://doi.org/10.5281/zenodo.5093192.
- Busch, Jonathan, Steinberger, Julia K., Dawson, David A., Purnell, Phil, Roelich, Katy, 2014. Managing Critical Materials with a Technology-Specific Stocks and Flows

Model. Environ. Sci. Technol. 48 (2), 1298–1305. https://doi.org/10.1021/es404877u.

- Calvo, Guiomar, Mudd, Gavin, Valero, Alicia, Valero, Antonio, 2016. Decreasing Ore Grades in Global Metallic Mining: A Theoretical Issue or a Global Reality? Resources 5 (4), 36. https://doi.org/10.3390/resources5040036.
- Calvo, Guiomar, Valero, Alicia, Valero, Antonio, 2018. Thermodynamic Approach to Evaluate the Criticality of Raw Materials and Its Application through a Material Flow Analysis in Europe: Evaluation of Critical Raw Materials Using Rarity. J. Ind. Ecol. 22 (4), 839–852. https://doi.org/10.1111/jiec.12624.
- Cambridge Econometrics. E3me technical manual v6. 1. Technical report, 2019. URL https://www.e3me.com/wp-content/uploads/2019/09/E3ME-Technical-Manual -v6.1-onlineSML.pdf. Accessed 16/12/2021.
- Capellán-Pérez, Iñigo, de Castro, Carlos, Gonzalez, Luis Javier Miguel, 2019. Dynamic Energy Return on Energy Investment (EROI) and material requirements in scenarios of global transition to renewable energies. Energ. Strat. Rev. 26, 39 100399. https:// doi.org/10.1016/j.esr.2019.100399.
- Capellán-Pérez, Iñigo, de Blas, Ignacio, Nieto, Jaime, de Castro, Carlos, Miguel, Luis Javier, Carpintero, Oscar, Mediavilla, Margarita, Lobejon, Luis Fernando, Ferreras-Alonso, Noelia, Rodrigo, Paula, Frechoso, Fernando, Alvarez-Antelo, David, 2020. MEDEAS: A new modeling framework integrating global biophysical and socioeconomic constraints. Energy Environ. Sci. 13 (3), 986–1017. https://doi.org/ 10.1039/C9EE02627D.
- Carmona, Luis, Whiting, Kai, Carrasco, Angeles, Sousa, Tania, Domingos, Tiago, 2017. Material Services with Both Eyes Wide Open. Sustainability 9 (9), 1508. https://doi. org/10.3390/su9091508.
- Ceballos, Gerardo, Ehrlich, Paul R., Dirzo, Rodolfo, 2017. Biological annihilation via the ongoing sixth mass extinction signalled by vertebrate population losses and declines. Proc. National Acad. Sci. 114 (30), E6089–E6096. https://doi.org/10.1073/ pnas.1704949114.
- Chilean Copper Commission. Anuario de Estadisticas del Cobre y otros Minerales. Yearbook: copper and other minerals statistics. Technical report, Ministerio de Mineria, Gobierno de Chile, 2020. URL https://www.cochilco.cl/Paginas/Estadistica s/Publicaciones/Anuario.aspx. See reports for years 2010–2020. Last accessed 03/ 09/2021.
- Ciacci, Luca, Reck, Barbara K., Nassar, N.T., Graedel, T.E., 2015. Lost by Design. Environ. Sci. Tech. 49 (16), 1520–5851. https://doi.org/10.1021/es505515z.
- Ciacci, L., Fishman, T., Elshkaki, A., Graedel, T.E., Vassura, I., Passarini, F., 2020. Exploring future copper demand, recycling and associated greenhouse gas emissions in the EU-28. Glob. Environ. Chang. 63 https://doi.org/10.1016/j. gloenycha.2020.102093.
- Cook, Earl, 1976. Limits to Exploitation of Nonrenewable Resources. Last accessed: 12/ 07/2022 Science, New Series 191 (4228), 677–682. http://www.jstor.org/stable/ 1741483.
- Deetman, Sebastiaan, Pauliuk, Stefan, van Vuuren, Detlef P., van der Voet, Ester, Tukker, Arnold, 2018. Scenarios for Demand Growth of Metals in Electricity Generation Technologies, Cars, and Electronic Appliances. Environ. Sci. Technol. 52 (8), 4950–4959. https://doi.org/10.1021/acs.est.7b05549.
- Deetman, Sebastiaan, Marinova, Sylvia, van der Voet, Ester, van Vuuren, Detlef P., Edelenbosch, Oreane, Heijungs, Reinout, 2020. Modelling global material stocks and flows for residential and service sector buildings towards 2050. J. Clean. Prod. 245 https://doi.org/10.1016/j.jclepro.2019.118658.
- Deetman, S., de Boer, H.S., Van Engelenburg, M., van der Voet, E., van Vuuren, D.P., 2021. Projected material requirements for the global electricity infrastructure – generation, transmission and storage. Resour. Conserv. Recycl. 164 https://doi.org/ 10.1016/j.resconrec.2020.105200.
- de Koning, Arjan, Kleijn, Rene, Huppes, Gjalt, Sprecher, Benjamin, van Engelen, Guus, Tukker, Arnold, 2018. Metal supply constraints for a low-carbon economy? Resour. Conserv. Recycl. 129, 202–208. https://doi.org/10.1016/j.resconrec.2017.10.040.

- Despres, J., Keramidas, K., Schmitz, A., Schade, B., Diaz Vazquez, A., Mima, S., Russ, H., Wiesenthal, T., 2018. POLES-JRC model documentation: 2018 update. 2018. https://doi.org/10.2760/814959.
- Dong, Di, van Oers, Lauran, Tukker, Arnold, van der Voet, Ester, 2020. Assessing the future environmental impacts of copper production in China: Implications of the energy transition. J. Clean. Prod. 274 https://doi.org/10.1016/j. iclepro.2020.122825.
- E3Modelling. GEM-E3. Model Manual. Technical report, 2017. URL https://e3modelling. com/modelling-tools/gem-e3/. Last accessed: 12/07/2022.
- E3Modelling. PRIMES Model. Detailed model description. Version 2018. Technical report, 2018. URL https://e3modelling.com/modelling-tools/primes/. Last accessed: 12/07/2022.
- EIA, 2021. World Energy Projection System (WEPS) Module Documentation. Technical report, Last accessed: 12/07/2022. U.S. Energy Information Administration.
- El Rasafi, Taoufik, Nouri, Mohamed, Haddioui, Abdelmajid, 2021. Metals in mine wastes: environmental pollution and soil remediation approaches – a review. Geosyst. Eng. 24 (3), 157–172. https://doi.org/10.1080/12269328.2017.1400474.
- Elshkaki, Ayman, Shen, Lei, 2019. Energy-material nexus: The impacts of national and international energy scenarios on critical metals use in China up to 2050 and their global implications. Energy 180, 903–917. https://doi.org/10.1016/j. energy 2019 05 156
- Elshkaki, T.E. Ayman, Graedel, Luca Ciacci, Reck, Barbara K., 2016. Copper demand, supply, and associated energy use to 2050. Global Environ. Change 39, 305–315. https://doi.org/10.1016/j.gloenvcha.2016.06.006.
- Elshkaki, T.E. Ayman, Graedel, Luca Ciacci, Reck, Barbara K., 2018. Resource Demand Scenarios for the Major Metals. Environ. Sci. Technol. 52 (5), 2491–2497. https:// doi.org/10.1021/acs.est.7b05154.
- Entwistle, Jane A., Hursthouse, Andrew S., Marinho Reis, Paula A., Stewart, Alex G., 2019. Metalliferous Mine Dust: Human Health Impacts and the Potential Determinants of Disease in Mining Communities. Curr. Pollut. Rep. 5 (3), 67–83. https://doi.org/10.1007/s40726-019-00108-5.
- Fizaine, Florian, Court, Victor, 2015. Renewable electricity producing technologies and metal depletion: A sensitivity analysis using the EROI. Ecol. Econ. 110 https://doi. org/10.1016/j.ecolecon.2014.12.001.
- Fricko, Oliver Fricko, Petr Havlik, Joeri Rogelj, Zbigniew Klimont, Mykola Gusti, Nils Johnson, Peter Kolp, Manfred Strubegger, Hugo Valin, Markus Amann, Tatiana Ermolieva, Nicklas Forsell, Mario Herrero, Chris Heyes, Georg Kindermann, Volker Krey, David L. McCollum, Michael Obersteiner, Shonali Pachauri, Shilpa Rao, Erwin Schmid, Wolfgang Schoepp, and Keywan Riahi. The marker quantification of the Shared Socioeconomic Pathway 2: A middle-of-the-road scenario for the 21st century. 42:251–267, 2017. https://doi.org/10.1016/j.gloenvcha.2016.06.004.
- Fujimori, Shinichiro, Hasegawa, Tomoko, Masui, Toshihiko, Takahashi, Kiyoshi, Herran, Diego Silva, Dai, Hancheng, Hijioka, Yasuaki, Kainuma, Mikiko, 2017. SSP3: AIM implementation of Shared Socioeconomic Pathways. Glob. Environ. Chang. 42, 268–283. https://doi.org/10.1016/j.gloenvcha.2016.06.009.
- Fujimori, Shinichiro, Hasegawa, Tomoko, Masui, Toshihiko, 2017b. AIM/CGE V2.0: Basic Feature of the Model. In: Fujimori, Shinichiro, Kainuma, Mikiko, Masui, Toshihiko (Eds.), Post-2020 Climate Action. Springer Singapore, Singapore, p. 13. https://doi.org/10.1007/978-981-10-3869-3.
- Glaister, Bonnie J., Mudd, Gavin M., 2010. The environmental costs of platinum–PGM mining and sustainability: Is the glass half-full or half-empty? Miner. Eng. 13. https://doi.org/10.1016/j.mineng.2009.12.007.
- Graedel, T.E., Allwood, Julian, Birat, Jean-Pierre, Buchert, Matthias, Hageluken, Christian, Reck, Barbara K., Sibley, Scott F., Sonnemann, Guido, 2011. United Nations Environment Programme, and Working Group on the Global Metal Flows. Recycling rates of metals: a status report. Last accessed: 12/07/2022 International Resource Panel, United Nations Environmental Program. https://www. unep.org/resources/report/recycling-rates-metals-status-report.
- Grancea, Luminita, Hanly, Adrienne, 2018. Uranium 2018: Resources, Production and Demand. Technical report, Last accessed: 12/07/2022. Organisation for Economic Co-Operation and Development.
- Grubler, Arnulf, Wilson, Charlie, Bento, Nuno, Boza-Kiss, Benigna, Krey, Volker, McCollum, David L., Rao, Narasimha D., Riahi, Keywan, Rogelj, Joeri, De Stercke, Simon, Cullen, Jonathan, Frank, Stefan, Fricko, Oliver, Guo, Fei, Gidden, Matt, Havlık, Petr, Huppmann, Daniel, Kiesewetter, Gregor, Rafaj, Peter, Schoepp, Wolfgang, Valin, Hugo, 2018. A low energy demand scenario for meeting the 1.5 °C target and sustainable development goals without negative emission technologies. Nature Energy 3 (6), 515–527. https://doi.org/10.1038/s41560-018-0172-6.
- Haas, Willi, Krausmann, Fridolin, Wiedenhofer, Dominik, Heinz, Markus, 2015. How Circular is the Global Economy?: An Assessment of Material Flows, Waste Production, and Recycling in the European Union and the World in 2005: How Circular is the Global Economy? J. Ind. Ecol. 19 (5), 765–777. https://doi.org/ 10.1111/jiec.12244.
- Haberl, Helmut, Wiedenhofer, Dominik, Virag, Doris, Kalt, Gerald, Plank, Barbara, Brockway, Paul, Fishman, Tomer, Hausknost, Daniel, Krausmann, Fridolin, Leon-Gruchalski, Bartholomaus, 2020. A systematic review of the evidence on decoupling of GDP, resource use and GHG emissions, part II: synthesizing the insights. Environ. Res. Lett. 15 (6) https://doi.org/10.1088/1748-9326/ab842a.
- Hammond, Geoff, Jones, Craig, 2011. Inventory of Carbon and Energy (ICE) version 2.0. University of Bath. Technical report.
- Haque, N., Norgate, T., 2015. Life cycle assessment of iron ore mining and processing. Iron Ore 615–630. https://doi.org/10.1016/B978-1-78242-156-6.00020-4.
- Harmsen, J.H.M., Roes, A.L., Patel, M.K., 2013. The impact of copper scarcity on the efficiency of 2050 global renewable energy scenarios. Energy 50, 62–73. https://doi. org/10.1016/j.energy.2012.12.006.

- Heck, Vera, Gerten, Dieter, Lucht, Wolfgang, Popp, Alexander, 2018. Biomass-based negative emissions difficult to reconcile with planetary boundaries. Nature Climate Change 8 (2), 151–155. https://doi.org/10.1038/s41558-017-0064-y.
- Heun, Matthew Kuperus, Owen, Anne, Brockway, Paul E., 2018. A physical supply-use table framework for energy analysis on the energy conversion chain. Appl. Energy 226, 1134–1162. https://doi.org/10.1016/j.apenergy.2018.05.109.
- Hickel, Jason, Kallis, Giorgos, 2019. Is Green Growth Possible? New Political Economy 1–18. https://doi.org/10.1080/13563467.2019.1598964.
- Hickel, Jason, Brockway, Paul, Kallis, Giorgos, Keyser, Lorenz, Lenzen, Manfred, Slamersak, Aljosa, Steinberger, Julia, Vorsatz, Diana Urge, 2021. Urgent need for post-growth climate mitigation scenarios. Nature Energy 6 (8), 766–768. https://doi. org/10.1038/s41560-021-00884-9.
- Holmberg, Kenneth, Kivikyto-Reponen, Paivi, Harkisaari, Pirita, Valtonen, Kati, Erdemir, Ali, 2017. Global energy consumption due to friction and wear in the mining industry. Tribol. Int. 115, 116–139. https://doi.org/10.1016/j. triboint.2017.05.010.
- Huppmann, D., Kriegler, E., Krey, V., Riahi, K., Rogelj, J., Rose, S.K., Weyant, J., Bauer, N., Bertram, C., Bosetti, V., Calvin, K., Doelman, J., Drouet, L., Emmerling, J., Frank, S., Fujimori, S., Gernaat, D., Grubler, A., Guivarch, C., Haigh, M., Holz, C., Iyer, G., Kato, E., Keramidas, K., Kitous, A., Leblanc, F., Liu, J.-Y., Eoffler, K., Luderer, G., Marcucci, A., McCollum, D., Mima, S., Popp, A., Sands, R.D., Sano, F., Streffer, J., Tsutsui, J., Van Vuuren, D., Vrontisi, Z., MarshallWise, Zhang, R., 2019. IAMC 1.5°C Scenario Explorer and Data hosted by IIASA. Integrated Assessment Modeling Consortium and International Institute for Applied Systems Analysis. URL https://data.ene.iiasa.ac.at/iamc-1.5c-explorer/#/login?redirect=%2Fworkspaces. Version 2.0. Last accessed: 12/07/2022.
- IEA. Energy Technology Perspectives 2017. Technical report, International Energy Agency, 2017. URL https://www.iea.org/reports/energy-technology-perspective s-2017. Last accessed: 12/07/2022.
- IEA. World Energy Balances 2019 Edition Database Documentation. Technical report, International Energy Agency, 2019.
- IEA, 2021. The Role of Critical Minerals in Clean Energy Transitions. Technical report, Last accessed: 12/07/2022. International Energy Agency, Paris.
- IEA. World Energy Model Documentation. Technical report, International Energy Agency, 2021.
- Kelly, T.D., Matos, G.R., 2016. Historical statistics for mineral and material commodities in the United States (2016 version). Series 140. Technical report, U.S. Geological Survey. URL https://www.usgs.gov/centers/nmic/historical-statistics-mineral-and -material-commodities-united-states. Last accessed: 12/07/2022.
- Keyser, Lorenz T., Lenzen, Manfred, 2021. 1.5 °C degrowth scenarios suggest the need for new mitigation pathways. Nat. Commun. 12 (1), 2676. https://doi.org/10.1038/ s41467-021-22884-9.
- Kleijn, Rene, van der Voet, Ester, Kramer, Gert Jan, van Oers, Lauran, van der Giesen, Coen, 2011. Metal requirements of low-carbon power generation. Energy 36 (9), 5640–5648. https://doi.org/10.1016/j.energy.2011.07.003.
  Koppelaar, R.H.E.M., Koppelaar, H., 2016. The Ore Grade and Depth Influence on Copper
- Koppelaar, R.H.E.M., Koppelaar, H., 2016. The Ore Grade and Depth Influence on Copper Energy Inputs. BioPhys. Econ. Resour. Qual. 1 (2), 11. https://doi.org/10.1007/ s41247-016-0012-x.
- Krausmann, Fridolin, Gingrich, Simone, Eisenmenger, Nina, Erb, Karl-Heinz, Haberl, Helmut, Fischer-Kowalski, Marina, 2009. Growth in global materials use GDP and population during the 20th century. Ecol. Econ. 68 (10), 2696–2705. https://doi.org/10.1016/j.ecolecon.2009.05.007.
- Krausmann, Fridolin, Lauk, Christian, Haas, Willi, Wiedenhofer, Dominik, 2018. From resource extraction to outflows of wastes and emissions: The socioeconomic metabolism of the global economy, 1900–2015. Glob. Environ. Chang. 52, 131–140. https://doi.org/10.1016/j.gloenvcha.2018.07.003.

Krey, V., Havlik, P., Kishimoto, P.N., Fricko, O., Zilliacus, J., Gidden, M., Strubegger, M., Kartasasmita, G., Ermolieva, T., Forsell, N., Gusti, M., Johnson, N., Kikstra, J., Kindermann, G., Kolp, P., Lovat, F., Mc, D.L., Min, J., Pachauri, S., 2021. MESSAGEix-GLOBIOM, release 2020. Technical report. Accessed 17/05/2020. International Institute for Applied Systems Analysis.

Kriegler, Kriegler, E., Bauer, N., Popp, A., Humperioder, F., Leimbach, M., Strefler, J., Baumstark, L., Leon Bodirsky, B., Hilaire, J., Klein, D., IoannaMouratiadou, IsabelleWeindl, Bertram, C., Dietrich, J.-P., Luderer, G., Pehl, M., Pietzcker, R., Piontek, F., Lotze Campen, H., Biewald, A., Bonsch, M., Giannousakis, A., Kreidenweis, U., Muller, C., Rolinski, S., Schultes, A., Schwanitz, J., Stevanovic, M., Calvin, K., Emmerling, J., Fujimori, S., Edenhofer, O., 2017. Fossil-fueled development (SSP5): An energy and resource intensive scenario for the 21st century. 42:297–315. https://doi.org/10.1016/j.gloenvcha.2016.05.015.

- Kuipers, Koen J.J., van Lauran, F.C.M., Oers, Verboon, Miranda, van der Voet, Ester, 2010. Assessing environmental implications associated with global copper demand and supply scenarios from to 2050. Glob. Environ. Chang. 49 (106–115), 2018. https://doi.org/10.1016/j.gloenvcha.2018.02.008.
- Lahiri-Dutt, Kuntala, 2018. Between the Plough and the Pick: Informal, Artisanal and Small-Scale Mining in the Contemporary World. Last accessed: 12/07/2022. ANU Press.
- Lasky, S.G., 1950. How tonnage and grade relations help predict ore reserves. Eng. Mining J. 151 (4), 81–85.
- Le Treut, G., 2018. Methodological proposals for hybrid modelling: consequences for climate policy analysis in an open economy (France). URL https://hal.archives-ouver tes.fr/tel-01707559. Last accessed: 12/07/2022.
- Le Treut, G., 2020. Description of the IMACLIM-Country model: A country-scale computable general equilibrium model to assess macroeconomic impacts of climate policies. URL https://hal.archives-ouvertes.fr/hal-02949396. Last accessed: 12/07/ 2022.

#### E. Aramendia et al.

- Luckeneder, Sebastian, Giljum, Stefan, Schaffartzik, Anke, Maus, Victor, Tost, Michael, 2021. Surge in global metal mining threatens vulnerable ecosystems. Glob. Environ. Chang. 69 https://doi.org/10.1016/j.gloenvcha.2021.102303.
- Masson-Delmotte, V., Zhai, P., Portner, H.O., Roberts, D., Skea, J., Shukla, P.R., Pirani, A., Moufouma-Okia, W., Pean, C., Pidcock, R., Matthews, J.B.R., Chen, Y., Zhou, X., Gomis, M.I., Lonnoy, E., Maycock, T., Tignor, M., Waterfields, T., 2018. Global Warming of 1.5oC. An IPCC Special Report on the impacts of global warning of 1.5oC above pre-industrial levels and related global greenhouse gas emission pathways, in the context of strengthening the global response to the threat of climate change, sustainable development, and efforts to eradicate poverty. Technical report, IPCC.
- Masson-Delmotte, V., Zhai, P., Connors, S.L., Pean, C., Berger, S., Caud, N., Chen, Y., Goldfarb, L., Gomis, M.I., Huang, M., Leitzell, K., Lonnoy, E., Matthews, J.B.R., Maycock, T.K., Waterfield, T., Yelekci, O., Yu, R., Zhou, B., 2021. Summary for Policymakers. Climate Change 2021: The Physical Science Basis. Contribution of Working Group I to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change. IPCC, Cambridge University Press. Technical report.
- Moreau, Vincent, Dos Reis, Piero, Vuille, Francois, 2019. Enough Metals? Resource Constraints to Supply a Fully Renewable Energy System. Resources 8 (1), 29. https://doi.org/10.3390/resources8010029.
- Morrell, Stephen, 2004. An alternative energy-size relationship to that proposed by Bond for the design and optimisation of grinding circuits. Int. J. Miner. Process. 9 https:// doi.org/10.1016/j.minpro.2003.10.002.
- Moss, R.L., Tzimas, Evangelos, Willis, Peter, Josie Arendorf, P., Thompson, A., Chapman, N., Morley, E., Sims, R Bryson, Peason, J., 2013. Critical Metals in the Path towards the Decarbonisation of the EU Energy Sector. Technical report, Last accessed 12/07/2022. Joint Research Center, European Comission.
- Mudd, Gavin M., 2007. Global trends in gold mining: Towards quantifying environmental and resource sustainability. Resour. Policy 32 (1–2), 42–56. https:// doi.org/10.1016/j.resourpol.2007.05.002.
- Mudd, Gavin M., 2010. The Environmental sustainability of mining in Australia: key mega-trends and looming constraints. Resour. Policy 35 (2), 98–115. https://doi. org/10.1016/j.resourpol.2009.12.001.
- Mudd, Gavin M., 2010. Global trends and environmental issues in nickel mining: Sulfides versus laterites. Ore Geol. Rev. 38 (1–2), 9–26. https://doi.org/10.1016/j. oregeorev.2010.05.003.
- Mudd, Gavin M., 2012. Sustainability Reporting and the Platinum Group Metals: A Global Mining Industry Leader? Platinum Metals Re. 56 (1), 2–19. https://doi.org/ 10.1595/147106711X614713.
- Mudd, Gavin M., Jowitt, Simon M., Werner, Timothy T., 2017. The world's lead-zinc mineral resources: Scarcity, data, issues and opportunities. Ore Geol. Rev. 80, 1160–1190. https://doi.org/10.1016/j.oregeorev.2016.08.010.
- Muller, J., Frimmel, H.E., 2010. Numerical Analysis of Historic Gold Production Cycles and Implications for Future Sub-Cycles. Open Geol. J. 4 (1), 29–34. https://doi.org/ 10.2174/1874262901004010029.
- Norgate, T., Haque, N., 2010. Energy and greenhouse gas impacts of mining and mineral processing operations. J. Clean. Prod. 18 (3), 266–274. https://doi.org/10.1016/j. jclepro.2009.09.020.
- Norgate, Terry, Haque, Nawshad, 2012. Using life cycle assessment to evaluate some environmental impacts of gold production. J. Clean. Prod. 29–30, 53–63. https:// doi.org/10.1016/j.jclepro.2012.01.042.
- Norgate, Terry, Haque, Nawshad, 2012. Using life cycle assessment to evaluate some environmental impacts of gold production. J. Clean. Prod. 29–30, 53–63. https:// doi.org/10.1016/j.jclepro.2012.01.042.
- Norgate, T., Jahanshahi, S., 2010. Low grade ores Smelt, leach or concentrate? Miner. Eng. 23 (2), 65–73. https://doi.org/10.1016/j.mineng.2009.10.002.
- Norgate, Terry, Jahanshahi, Sharif, 2011. Reducing the greenhouse gas footprint of primary metal production: Where should the focus be? Miner. Eng. 24 (14), 1563–1570. https://doi.org/10.1016/j.mineng.2011.08.007.
- Norgate, T., Jahanshahi, S., 2011. Assessing the energy and greenhouse gas footprints of nickel laterite processing. Miner. Eng. 24 (7), 698–707. https://doi.org/10.1016/j. mineng.2010.10.002.
- Norgate, T.E., Jahanshahi, S., Rankin, W.J., 2007. Assessing the environmental impact of metal production processes. J. Clean. Prod. 15 (8–9), 838–848. https://doi.org/ 10.1016/j.jclepro.2006.06.018.
- Norgate, T., Haque, N., Koltun, P., 2014. The impact of uranium ore grade on the greenhouse gas footprint of nuclear power. J. Clean. Prod. 84, 360–367. https://doi. org/10.1016/j.jclepro.2013.11.034.
- Northey, S., 2013. Using sustainability reporting to assess the environmental footprint of copper mining. J. Clean. Prod. 11 https://doi.org/10.1016/j.jclepro.2012.09.027.
- Northey, S., Mohr, S., Mudd, G.M., Weng, Z., Giurco, D., 2014. Modelling future copper ore grade decline based on a detailed assessment of copper resources and mining. Resour. Conserv. Recycl. 83, 190–201. https://doi.org/10.1016/j. resconrec.2013.10.005.
- Nuss, Philip, Eckelman, Matthew J., 2014. Life Cycle Assessment of Metals: A Scientific Synthesis. PLoS One 9 (7). https://doi.org/10.1371/journal.pone.0101298.
- O'Neill, Brian C., Kriegler, E., Ebi, K.L., Kemp Benedict, E., Riahi, K., Rothman, D.S., Van Ruijven, B.J., Van Vuuren, D.P., Birkmann, J., Kok, K., Levy, M., Solecki, W., 2017. The roads ahead: Narratives for shared socioeconomic pathways describing world futures in the 21st century. 42, 169–180. https://doi.org/10.1016/j.gloenvcha.20 15.01.004.

Page, Norman J., Creasey, S.C., Park, Menlo, 1975. Ore grade, metal production, and energy. J. Res. US Geological Survey 6.

Paoli, Leonardo, Cullen, Jonathan, 2020. Technical limits for energy conversion efficiency. Energy 192. https://doi.org/10.1016/j.energy.2019.116228.

- Parker, David J., Mnaughton, Cameron S., Sparks, Gordon A., 2016. Life Cycle Greenhouse Gas Emissions from Uranium Mining and Milling in Canada. Environ. Sci. Technol.gy 50 (17), 9746–9753. https://doi.org/10.1021/acs.est.5b06072.
- Parrique, Timothee, Barth, Jonathan, Briens, Francois, Kerschner, Christian, Kraus-Polk, Alejo, 2019. Decoupling Debunked. Last accessed 12/07/2022 European Environmental Bureau. https://eeb.org/library/decoupling-debunked/.
- Pauliuk, Stefan, Milford, Rachel L., Muller, Daniel B., Allwood, Julian M., 2013. The Steel Scrap Age. Environ. Sci. Technol. 47 (7), 3448–3454. https://doi.org/10.1021/ es303149z.
- Pigneur, J., 2019. Mise au point d'une méthode intégrée dánalyse des impacts des filiéres de matiéres premiéres minérales. PhD thesis. URL https://tel.archives-ouvertes.fr/tel -03123793. Last accessed: 12/07/2022.
- Pothen, Frank, 2017. A structural decomposition of global Raw Material Consumption. Ecol. Econ. 141, 154–165. https://doi.org/10.1016/j.ecolecon.2017.05.032.
- Prior, T., Giurco, D., Mudd, G., Mason, L., Behrisch, J., 2012. Resource depletion, peak minerals and the implications for sustainable resource management. Glob. Environ. Chang. 22 (3), 577–587. https://doi.org/10.1016/j.gloenvcha.2011.08.009.
- Rankin, W. John, 2011. Minerals, metals and sustainability: meeting future material needs. CSIRO Publishing. ISBN 0-643-09726-0.
- Rogelj, Joeri, den Elzen, Michel, Hohne, Niklas, Fransen, Taryn, Fekete, Hanna, Winkler, Harald, Roberto Schaeffer, Fu, Sha, Keywan Riahi, Meinshausen, Malte, 2016. Paris Agreement climate proposals need a boost to keep warming well below 2 oC. Nature 534 (7609), 631–639. https://doi.org/10.1038/nature18307.
- Rotzer, Nadine, Schmidt, Mario, 2020. Historical, Current, and Future Energy Demand from Global Copper Production and Its Impact on Climate Change. Resources 9 (4), 44. https://doi.org/10.3390/resources9040044.
- Schafer, Philipp, Schmidt, Mario, 2020. Discrete-Point Analysis of the Energy Demand of Primary versus Secondary Metal Production. Environ. Sci. Technol. 54 (1), 507–516. https://doi.org/10.1021/acs.est.9b05101.
- Schleussner, Carl-Friedrich, Rogelj, Joeri, Schaeffer, Michiel, Lissner, Tabea, Licker, Rachel, Fischer, Erich M., Knutti, Reto, Levermann, Anders, Frieler, Katja, Hare, William, 2016. Science and policy characteristics of the Paris Agreement temperature goal. Nature Climate Change 6 (9), 827–835. https://doi.org/10.1038/ nclimate3096.

#### Shell. Shell World Energy Model. A view to 2100. Technical report, 2017.

- Skinner, Brian, 1976. A Second Iron Age Ahead? The distribution of chemical elements in the earth's crust sets natural limits to man's supply of metals that are much more important to the future of society than limits on energy. American Scientist 13. Last accessed: 12/07/2022.
- Smith, Pete, Davis, Steven J., Creutzig, Felix, Fuss, Sabine, Minx, Jan, Gabrielle, Benoit, Kato, Etsushi, Jackson, Robert B., Cowie, Annette, Kriegler, Elmar, van Vuuren, Detlef P., Rogelj, Joeri, Ciais, Philippe, Milne, Jennifer, Canadell, Josep G., McCollum, David, Peters, Glen, Andrew, Robbie, Krey, Volker, Shrestha, Gyami, Friedlingstein, Pierre, Gasser, Thomas, Grubler, Arnulf, Heidug, Wolfgang K., Jonas, Matthias, Jones, Chris D., Kraxner, Florian, Littleton, Emma, Lowe, Jason, Moreira, Jose Roberto, Nakicenovic, Nebojsa, Obersteiner, Michael, Patwardhan, Anand, Mathis Rogner, Ed, Rubin, Ayyoob Sharifi, Torvanger, Asbjorn, Yamagata, Yoshiki, Edmonds, Jae, Yongsung, Cho, 2016. Biophysical and economic limit to accenting CO2 privinger, Netron Charge, Chy 42–50. https://doi.org/10.1016/j.jpii.0016.
- limits to negative CO2 emissions. Nature Climate Change 6 (1), 42–50. https://doi.org/10.1038/nclimate2870.
   Smith, Christopher J., Forster, Piers M., Allen, Myles, Fuglestvedt, Jan, Millar, Richard J., Rogelj, Joeri, Zickfeld, Kirsten, 2019. 38 Current fossil fuel infrastructure does not
- Rogelj, Joeri, Zickfeld, Kirsten, 2019. 38 Current fossil fuel infrastructure does not yet commit us to 1.5° C warming. Nature Commun. 10 (1), 101. https://doi.org/10.1038/s41467-018-07999-w.
- Sonter, Laura J., Dade, Marie C., Watson, James E.M., Valenta, Rick K., 2020. Renewable energy production will exacerbate mining threats to biodiversity. Nat. Commun. 11 (1), 4174. https://doi.org/10.1038/s41467-020-17928-5.
- Steffen, Will, Broadgate, Wendy, Deutsch, Lisa, Gaffney, Owen, Ludwig, Cornelia, 2015. The trajectory of the Anthro pocene: The Great Acceleration. The Anthropocene Review 2 (1), 81–98. https://doi.org/10.1177/2053019614564785, 2053-020X.
- Steffen, W., Richardson, K., Rockstrom, J., Cornell, S.E., Fetzer, I., Bennett, E.M., Biggs, R., Carpenter, S.R., de Vries, W., de Wit, C.A., Folke, C., Gerten, D., Heinke, J., Mace, G.M., Persson, L.M., Ramanathan, V., Reyers, B., Sorlin, S., 2015. Planetary boundaries: Guiding human development on a changing planet. Science 347 (6223), 1259855. https://doi.org/10.1126/science.1259855.
- Stehfest, Elke, van Vuuren, Detlef, Kram, Tom, Bouwman, Lew, Alkemade, Rob, Bakkenes, Michel, Biemans, Hester, Bouwman, Arno, den Elzen, Michel, Janse, Jan, Lucas, Paul, van Minnen, Jelle, Muller, Christoph, Prins, Anne Gerdien, 2014. Integrated Assessment of Global Environmental Change with IMAGE 3.0. Model description and policy applications. Technical report, Last accessed: 12/07/2022. PBL Netherlands Environmental Agency.
- Steinberger, Julia K., Krausmann, F., Getzner, M., Schandl, H., West, J., 2013. Development and Dematerialization: An International Study. 8:e70385. https://doi. org/10.1371/journal.pone.0070385.
- Stephens, Carolyn, Ahern, Mike, 2001. Worker and community health impacts related to mining operations internationally: A rapid review of the literature. Last accessed: 24/08/21. London School of Hygiene & Tropical Medicine, London.
- Stern, David I., Kander, Astrid, 2012. The Role of Energy in the Industrial Revolution and Modern Economic Growth. Energy J. 33 (3) https://doi.org/10.5547/ 01956574.33.3.5.
- Topp, V., 2008. Australia, and Productivity Commission. Productivity in the mining industry: measurement and interpretation: staff working paper. Productivity Commission, Melbourne. ISBN 978-1-74037-271-8. URL https://www.pc.gov.au/re search/supporting/mining-productivity/mining-productivity.pdf. Last accessed: 12/ 07/2022.

#### E. Aramendia et al.

- United Nations editor. International Standard Industrial Classification of All Economic Activities (ISIC). Number no. 4, rev. 4 in Statistical Papers. Series M. United Nations, New York, rev. 4 edition, 2008. ISBN 978-92-1-161518-0. URL: https://unstats.un. org/unsd/classifications/Econ/Download/In%20Text/ISIC\_Rev\_4\_publication\_Engli sh.pdf. Last, accessed: 12/07/2022.
- US-DoE. Mining industry energy bandwidth study. Technical report, United States Department of Energy, 2007. URL https://www.energy.gov/sites/prod/files/20 13/11/f4/mining\_bandwidth.pdf. Accessed 17/12/2021.
- Valero, Alicia, Valero, Antonio, Calvo, Guiomar, Ortego, Abel, Ascaso, Sonia, Palacios, Jose-Luis, 2018. Global material requirements for the energy transition. An exergy flow analysis of decarbonisation pathways. Energy 159, 1175–1184. https:// doi.org/10.1016/j.energy.2018.06.149.
- Valero, Alicia, Valero, Antonio, Calvo, Guiomar, Ortego, Abel, 2018. Material bottlenecks in the future development of green technologies. Renew. Sustain. Energy Rev. 93, 178–200. https://doi.org/10.1016/j.rser.2018.05.041.
- Van der Voet, Ester, Van Oers, Lauran, Verboon, Miranda, Kuipers, Koen, 2019. Environmental Implications of Future Demand Scenarios for Metals: Methodology and Application to the Case of Seven Major Metals. J. Ind. Ecol. 23 (1), 141–155. https://doi.org/10.1111/jiec.12722.
- van Ruijven, Bas J., van Vuuren, Detlef P., Boskaljon, Willem, Neelis, Maarten L., Saygin, Deger, Patel, Martin K., 2016. Longterm model-based projections of energy use and CO2 emissions from the global steel and cement industries. Resour. Conserv. Recycl. 112 https://doi.org/10.1016/j.resconrec.2016.04.016.

- van Vuuren, Detlef P., Hof, Andries F., van Sluisveld, Mariesse A.E., Riahi, Keywan, 2017. Open discussion of negative emissions is urgently needed. Nature. Energy 2 (12), 902–904. https://doi.org/10.1038/s41560-017-0055-2.
- Vaughan, Naomi E., Gough, Clair, 2016. Expert assessment concludes negative emissions scenarios may not deliver. Environ. Res. Lett. 11 (9) https://doi.org/10.1088/1748-9326/11/9/095003.
- Vidal, Olivier, Le Boulzec, Hugo, Andrieu, Baptiste, Verzier, Francois, 2021. Modelling the Demand and Access of Mineral Resources in a Changing World. Sustainability 14 (1), 11. https://doi.org/10.3390/su14010011.
- Viebahn, Peter, 2015. Assessing the need for critical minerals to shift the German energy system towards a high proportion of renewables. Renew. Sustain. Energy Rev. page 17. https://doi.org/10.1016/j.rser.2015.04.070.
- Watari, Takuma, 2020. Review of critical metal dynamics to 2050 for 48 elements. Resour., Conserv. Recycl. 17 https://doi.org/10.1016/j.resconrec.2019.104669
- Watari, Takuma, McLellan, Benjamin, Ogata, Seiichi, Tezuka, Tetsuo, 2018. Analysis of Potential for Critical Metal Resource Constraints in the International Energy Agency's Long-Term Low-Carbon Energy Scenarios. Minerals 8 (4), 156. https://doi. org/10.3390/min8040156.
- Weng, Zhehan, Haque, Nawshad, Mudd, Gavin M., Jowitt, Simon M., 2016. Assessing the energy requirements and global warming potential of the production of rare earth elements. J. Clean. Prod. 139, 1282–1297. https://doi.org/10.1016/j. iclenro.2016.08.132.
- Wiedmann, Thomas O., Schandl, Heinz, Lenzen, Manfred, Moran, Daniel, Suh, Sangwon, West, James, Kanemoto, Keiichiro, 2015. The material footprint of nations. Proc. Natl. Acad. Sci. 112 (20), 6271–6276. https://doi.org/10.1073/pnas.1220362110.